Journal of Cleaner Production 172 (2018) 2999-3018

Contents lists available at ScienceDirect

Journal of Cleaner Production

journal homepage: www.elsevier.com/locate/jclepro

Eco-innovation in the transition to a circular economy: An analytical literature review



^a CENSE – Center for Environmental and Sustainability Research, NOVA School of Science and Technology, NOVA University, Lisbon, Portugal ^b Instituto Universitário de Lisboa (ISCTE-IUL), Business Research Unit (BRU-IUL), Lisbon, Portugal

^c UECE – Research Unit on Complexity and Economics, ISEG/UL - University of Lisbon, Lisbon, Portugal

^d SPRU, University of Sussex, UK

ARTICLE INFO

Article history: Received 13 June 2017 Received in revised form 2 October 2017 Accepted 15 November 2017 Available online 15 November 2017

Keywords: Circular economy Eco-innovation Sustainability transition Sustainable development Clean congruence

ABSTRACT

In the ongoing sustainability debate, the circular economy (CE) has been steadily gaining ground as a new paradigm. At the same time, eco-innovation (EI) has been recognised as a key element in carrying out the transition from a linear to a circular system of production and consumption. However, little information can be found concerning whether and how EI can actually facilitate the change to a CE. While extensive literature on EI, and a growing body of research exploring the CE, already exist, there is, as yet, no comprehensive understanding concerning the connections between these two concepts. Drawing on academic contributions from the fields of EI and CE, this analysis seeks to clarify and synthesise findings at the intersection of these two fields. The aim is threefold: derive literature-based working definitions of CE and EI; review the role of EI at CE's macro, meso, and micro levels; and characterise CE-inducing EI in terms of targets, mechanisms and impacts. Our literature review shows that an El-driven techno-economic transition to a CE requires specific solutions, i.e. different forms of El-driven "clean congruence" at distinct levels of operation. Generally speaking, movement toward a CE is found to be contingent on "systemic" EI, that is, not only intense in technology but also involving dynamic and holistic combinations of service innovations and novel organisational set-ups.

© 2017 Elsevier Ltd. All rights reserved.

1. Introduction

Continued human use and abuse of natural resources is pushing global ecosystems to the brink. Several global tipping points have already been reached, increasing the risk of cascading irreversible environmental changes (Rockström et al., 2009). Recent decades have highlighted the importance of decoupling economic growth and social development from resource exploitation and waste. One of the defining challenges of the 21st century seems to be how to accommodate economic development among competing countries, and the continuous rise of living standards of a world population estimated to reach 10 billion by 2050, in a context of limited natural resources without jeopardising the sustainability of the global environment (OECD, 2012).

* Corresponding author.

In the light of the limitations of the conventional economy, a more circular approach is gaining traction. A view referred to as the "circular economy" (CE) has been put forward as a strategic approach, placing closed-loop thinking at the heart of businesses, industrial organisation and national agendas (see, e.g. Preston, 2012). The CE is a concept inspired on natural ecosystems and postulates moving away from a notion of a linear economy (based on unidirectional extraction, production, distribution, consumption and disposal activities) towards a permanently regenerative economy in an effort to rethink all of a product's life cycle. The CE focuses on the design of processes and products aiming to minimize negative environment and societal impacts, reducing the use of non-renewable resources, eliminating toxic and hazardous materials, and increasing product lifespan, as well as maximising the potential for reusing products and recovering materials (IAU, 2013). It proposes models for value creation that support sustainable economic development, through loops of reuse, restoration and renewability, where waste is residual or converted into an input into other processes thus shifting the emphasis to the provision of





Cleane Productio

E-mail addresses: aij.silva@fct.unl.pt (A. de Jesus), mpa@fct.unl.pt (P. Antunes), rfs@fct.unl.pt (R. Santos), sfm@iscte.pt (S. Mendonça).

functionality and "service" rather than ownership and material production (EMF, 2012; Stahel and Reday-Mulvey, 1981).

Building on early definitions CE ideas have gained additional relevance as a research topic over the last decade (Andersen, 2007), driven by organisations such as the United Nations (UNEP, 2014) and the European Union (EC, 2015) as well as the work of private agents as the Ellen MacArthur Foundation (EMF, 2012, 2013, 2014). Nevertheless, "while businesses and governments are recognising the need for change, there is confusion on what needs to be changed and how it can best be accomplished." (Schulte, 2013, p. 47).

Meanwhile, eco-innovation (EI) has been emphasised as a core driver for change in the transition to sustainability (Kemp, 2010). It is defined as innovation, in all of its forms (product, process, marketing, organisational - see OECD, 2005), yielding both ecological and economic gains (Carrillo-Hermosilla et al., 2010). In other words, the concept has been recognised as a key element in the development of competitive technologies and institutional forms (including new business models) that allow "environmental benefits", including greater efficiency in consumption and use of resources (EC, 2012). In the policy arena, EI has been called "a catalyst" of a CE (Potočnik, 2014) and a key component in the transition from a linear to a circular system of production and consumption (EIO, 2016).

However, an analysis of the intersection of the CE and El concepts seems lacking, with few studies considering the explicit importance of El towards a CE (ElO, 2016). If aligning innovation activities with a more sustainable path is a central requirement for a techno-economic paradigm shift (Mirata and Emtairah, 2005), how can the innovation agenda be geared towards a CE? What changes are instrumental for such a structural break? Can the concept of El bring about the changes required for the CE framework to be deployed and reinforced?

The way in which EI is to drive the pro-CE transition remains an insufficiently addressed issue. Using an innovation studies perspective this paper aims to contribute to the sustainability transition debate. More comprehensive research seem instrumental to understand such a transformative transition, driven by dynamic and holistic CE-friendly business models and public policies. Understanding the role of EI will help actors and institutions to better adjust and calibrate their CE efforts. Business actors, in particular, would benefit from this analysis, so as to be able to both redesign and pursue sustainable business models from the outset. As for policy makers, an integrated understanding of EI, and its relationship to CE, could underpin initiatives that take uncertainty and feedback loops into account. On the basis of a conceptuallydriven literature review we propose the concept of, to mint a term, "clean congruence" as a link between the CE and EI bodies of work.

This paper seeks to provide a view of the role of El within a CE framework by drawing on overlapping academic contributions from the fields of El and the CE and by appraising existing insights at the point where these agendas intersect. The three main objectives of this paper are thus to: 1) derive literature-based working definitions and characteristics of El and CE; 2) review and assess the relationship between the different dimensions of El and the various levels of a CE, and; 3) generate an overview of the types of El that may be most instrumental in achieving a CE.

The paper is organised as follows: Section 2 focuses on conceptual features and highlights the little charted common ground between the concepts of EI and the CE. Section 3 introduces the methodology, while Section 4 analyses the results stemming from the literature. Section 5 offers policy implications as structured by EI-CE connections. Finally, section 6 concludes by examining avenues for further research.

2. Eco-innovation and the circular economy

Even if one can intuitively argue that EI and the CE are closely related, and assume that achieving a CE without EI is unlikely, it remains to be seen in what specific ways this is so. Certainly not all EI is linked to a CE, and not all dimensions of CE require innovation. However, a zone of overlap is bound to exist. Therefore, in order to ascertain which innovations are more compatible with CE models, and how a CE is to be achieved through techno-economic change, a clearer understanding of the two concepts is useful.

2.1. Innovation dynamics in an evolving economy: the "proenvironment" family of eco-innovation related concepts

Since the seminal writings of Joseph Schumpeter (1928), it has been acknowledged that innovation is not just newness *per se*. It is, rather, a "new combination" of ideas and factors of production. Innovation is not only about technical sophistication but also about adaptation to a usage context, i.e. it is the introduction of an ingenious proposition into a specific, and sometimes quirky, economic and institutional setting (Fagerberg et al., 2004). That is to say, innovation is not simply science and technology. From this perspective, innovation is not understood to be automatic, it is neither a linear output from increased R&D, nor a passive reaction to market signals (Caraca et al., 2009).

Moreover, innovation is not necessarily better: novel outcomes are not inevitably superior to the status quo, from a welfare or sustainability point of view (Soete, 2013). What is technologically feasible is not necessarily ethically desirable or environmentally sound (UNEP, 2011). The 20th century mass-production technological regime was carbon-intensive and extraction-based, creating on hindsight fundamental questions about the meaning of the very notion of "progress". One implication is that innovation concepts may be liable to some revision, or even intellectual "creative destruction". As Schot and Kanger (2016), p. 25) stress, modifying "(...) the way we innovate (...)" is essential for transition. Transitions are complex dynamic processes involving a rich range of actors and discrete actions, and continued activities for a significant period of time, during which new products, services, business models and organisations emerge, either complementing or substituting incumbent ones, comprising an interacting sequence of technological and non-technological innovations (Markard et al., 2012; van den Bergh et al., 2011). As the environment became an area of prime policy concern, a cluster of concepts emerged concerning innovation focused on transition topics and broader societal challenges (Boons et al., 2013; Carrillo-Hermosilla et al., 2009; Rennings, 2000). This emerging "pro-environment" innovation agenda was beyond the scope of the industrial era (Freeman and Soete, 1997).

The entry and diffusion of an environmental angle of analysis into innovation studies has been characterised by some lexical variation. As innovation began to be conceived more and more as a dynamic process that evolves in real historical time and involves a multitude of different activities, not just formal R&D from a "hightech" supply-side but also shaping by the social and cultural environment (Castellaci et al., 2005; Balconi et al., 2010; Guan and Liu, 2016; Lee and Walsh, 2016), innovation studies benefited from the development of other fields of research such as sustainability and transition studies (Smith et al., 2010). Sustainability and transition studies emphasise science and technology as socially embedded processes; that is, knowledge is intertwined with mental maps, the expectations of consumers, is co-constructed by skills of users, and shaped by institutional/regulatory structures and infrastructures (Markard et al., 2012). There is little consensus on how to operationalise the approach to sustainability transitions; several viewpoints co-exist and a broad range of relevant theoretical approaches, encompassing several perspectives like evolutionary economic theory (Nelson and Winter, 1982), strategic niche management (Kemp et al., 1998), technological innovation systems (Bergek et al., 2008), multi-level perspective on sociotechnical transitions (Geels, 2002, 2011) or eco-innovation (Andersen, 2008; Kemp, 2010) just to enumerate a few.

Although terminological creativity can be taken as an early indicator of conceptual restlessness, there may be a point where "label proliferation" may hamper progress in a field (Alvarez et al., 2014; Barney, 2003). As Table 1 shows, terms emerging in the literature since the mid-1990s, linking innovation to environmental concerns, have somewhat distinct, yet related, definitions. "Environmental innovation", for example, is characterised as innovation with environmental benefits (van den Bergh et al., 2011; Weber and Hemmelskamp, 2005). By contrast, "Sustainable innovation" is thought of as more rounded innovation, addressing ecological, economic and social concerns, hence being more sensitive to the spatial, temporal and cultural context (Boons et al., 2013) and focusing not only on product and process innovations, but also on organisational models (Charter and Clark, 2007). In turn, "Green innovation" is described in terms of new or improved products and processes, with the aim of fostering environmental sustainability (Cuerva et al., 2014). More recently, "Business model innovation" seems also in line with this semantic field, being defined as innovation in the way organisations create, deliver and capture value, so as to maximise societal and environmental benefits (Bocken et al., 2014: Boons and Lüdeke-Freund. 2013).

As for "Eco-innovation" (EI), its initial "end of pipe" focus has recently been broadened in scope. EI is nowadays defined as a way of enabling economic performance that does not hinder sustainable development (i.e. economically, ecologically and socially sustainable performance) and is more positively defined, by the European Commission, as "resulting in or aiming at significant and demonstrable progress towards the goal of sustainable development, through reducing impacts on the environment, enhancing resilience to environmental pressures, or achieving a more efficient and responsible use of natural resources." (EC, 2011a, p. 2) El is also acknowledged as a way to increase competitiveness without (environmental and societal) negative impacts (OECD, 2009), and an indispensable condition for sustainability (Aghion et al., 2009; EC, 2011b). In spite of some irreducible variability, some efforts towards simplification and consolidation may be useful here. In this paper EI is taken as a streamlined and all-encompassing term for environmentally-sensitive innovation, and will be used preferentially. This term refers to all types of innovation addressing ecological concerns and/or having positive ecological effects (Jabbour et al., 2015). Considering the redirection of innovation studies towards "transformative innovation" (see Schot and Steinmueller, 2016), we take the overlap between EI and CE to be a fulcrum for realising the potential of a new clean and coherent techno-economic paradigm.

2.2. Untangling and re-focusing "eco-innovation"

For the purposes of policy-making, entrepreneurial decisionmaking and academic research, a clear definition of EI and its dimensions is helpful. A broad, but applicable, operational definition can be offered here as: *innovation that encompasses or results in environmental damage prevention, mitigation and recovery.* This definition (Silva and Mendonca in UN, 2015, p. 90) includes a number of critical aspects:

- improved environmental performance (i.e. green innovation);
- market efficient and clean results (i.e. environmental innovation);
- enduring and socially responsible benefits (i.e. sustainable innovation);
- holistic transformation (i.e. business model innovation for sustainability).

This definition provides a robust way of understanding the many different facets of EI, whilst also integrating the many diverse areas of analysis already undertaken on this issue. Drawing on existing EI typologies (OECD, 2010), inspired by the innovation guidelines of the Oslo Manual (OECD, 2009), EI is considered in its three main dimensions: key types of innovation (*targets*); the

Table 1

The family of "environmentally-friendly" concepts of innovation.

	Description	References
Environmental innovation	"() innovation can be beneficial to both the innovating firm and the environment".	Weber and Hemmelskamp, 2005, p. 3
Sustainable innovation	"Process where sustainable considerations (environmental, social, and financial) are integrated into company systems, from idea generation through to research, development and commercialization. This applies to products, services and technologies, as well as to new business and organisational models"; also "() adoption of new processes and systems at societal level".	Charter and Clarke, 2007, p. 9
	"() sustainable innovation brings into focus the relevance of () the relationships with other actors (i.e., suppliers and customers)".	Boons et al, 2013, p. 11
Green innovation	"() innovations in products, processes or business models lead the company to higher levels of environmental sustainability"	Cuerva et al. 2014, p. 104
Business model innovations for sustainability	Business model innovations for sustainability are defined as: "() innovations that create significant positive and/or significantly reduced negative impacts for the environment and/or society, through changes in the way the organisation, and its value-network, create, deliver and capture value (i.e. create economic value) or change their value propositions".	Bocken et al. 2014, p. 44
Eco-innovation	"() innovation which is fuelled by ecological issues ()" "() develop new ideas, behaviour, products and processes, apply or introduce them, which contribute to a reduction of environmental burdens or to ecologically specified sustainability targets". "() innovation that improves environmental performance ()"	Fussler and James, 1996, p. xi Rennings, 2000, p. 322 Carrillo-Hermosilla et al., 2010, p. 1075: 2009, p. 4
	"() the creation of new, or significantly improved, products (goods and services), processes, marketing methods, organisational structures and institutional arrangements which - with or without intent - lead to environmental improvements compared to relevant alternatives" "() any form of innovation resulting in or aiming at significant and demonstrable progress towards the goal of sustainable development"	OECD, 2010, p. 40 EC, 2011a, p. 2

nature of the change (*mechanisms*), and; resulting effects (*impacts*). Hence, and capitalising on this discussion, EI is operationally summed up as any innovation that: a) has positive environmental impacts, and; b) directly or indirectly avoids natural capital damage, while delivering cost efficiencies, market enhancement, or regulation considerations, and; c) results in new or improved goods and services, technological and non-technological processes, marketing or organisational schemes; d) is incremental or radical, and; e) involves an actor or a plurality of actors.

2.3. Transition to sustainability through CE-inducing approaches: the family of CE-friendly concepts

For its most part the global economy remains a system where activities, from tangible production to intangible contracts, routines and regulations, take place within a linear model of open-ended "take-make-dispose" resource exploitation. Notwithstanding the growing awareness that the use of the Earth's resources cannot be limitless, and the dissemination of related concepts, such as corporate social responsibility, this linear model remains essentially unchallenged (although future-oriented debates go back a long time, see Mendonça, 2017). Moreover, moving away from this model will not be an easy task as entrenched technical systems are made stiffer by risk avoidance and special interests with much to lose in the short run (Markard et al., 2012; Schulte, 2013).

In the post-Paris COP 21 context, expectations are high, with 175 governments (174 countries and the European Union) signing the initial agreement, originally with the United States and China among them (UNFCCC, 2016). However, various actors' interests do not align well, as the promotion of national economic competitiveness, in a fiercely dynamic global market, comes to terms with the impacts of continued environmental degradation. A new set-up may need to be based on "decoupling" development from resource consumption, by focusing on extended material life-cycles, reuse, re-manufacturing and recycling (UNEP, 2011). If the need for change is increasingly recognised, the specific pathways of transition remain much less defined. A number of perspectives for framing the discussion have been proposed in the literature, which have been instrumental in shaping the current understanding of the CE. Table 2 presents salient examples of these.

Several ideas behind the CE concept are not new practices. Animal waste by-products (e.g. pelts, blood and bones) have been used at least since Neolithic times in the making of other items, such as fabrics, shelters, weapons and jewellery (Desrochers, 2000). Similarly, even in the 19th century, the potential benefits arising from cooperative arrangements between manufacturers and consumers, through by-product exchange and service bartering, were already being enacted (Simmonds 1862; 1875 in Desrochers, 2000). The integrated concept of the CE emerged in the late 20th century, alongside concerns regarding planetary-level resource exhaustion; e.g. Boulding (1966) "spaceman economy" advocacy, which stressed the need to find a new balance in a "cyclical ecological system"; and Georgescu-Roegen's (1971) entropy approach to the economic system. CE as a label first appeared in Pearce and Turner (1990), discussed in a full chapter, where the case for the economic practicality of environmental values was developed, referring to the works of Boulding and Georgescu-Roegen, and arguing that natural systems also have waste but, unlike the traditional open-ended economy, they absorb and recycle it. The authors argued for "circular" material flows in the man-made economy. An economic system organised like nature, operating in loops, would reduce the need for new inputs, and delay the depletion of the "environment" (as a source of materials and a sink for waste). Resources should not simply end up as litter after usage, or as products that are simply designed to accommodate the next wave of supply; they should rather be transformed from one form to another, and converted back to new resources.

The notion of the CE eventually infused the field of "industrial ecology", especially in the USA, popularised by Robert Frosch and Nicholas E. Gallopoulos (1989) and Robert Ayres and Weaver (1998). Industrial ecology literature explicitly proposes the mimicking of natural systems' strategies as an industrial organisation template. It stresses the need for "material symbiosis" amongst different businesses and production processes, converting waste by-products into material inputs (Andersen, 2007, p. 133). In Europe, the industrial symbiosis concept has been taken up by many institutions and is widely used. The focus is on a "systems integration" view of companies exchanging by-products, closing each other's materials' cycles, and this is seen as an element that directly promotes CE implementation (Chertow, 2007; Lombardi and Laybourn, 2012).

In the late 1990's and early 2000's, critiques of traditional "industrial capitalism", which paradoxically both endangers the environment while also depending on it for natural resources, offered the notion of "natural capitalism" (Lovins et al., 1999). In this frame, environmental and economic benefits are based on more effective manufacturing processes, valuation, reuse and recycling of materials, in tune with CE considerations.

Other features of the CE can be found in the development of the "Cradle to Cradle" approach, and expected impacts on competitiveness, job creation, resource savings and waste prevention, emphasising the conversion of strict manufacturing into a nexus of self-feeding services (Stahel and Reday-Mulvey, 1981). This view was expanded over time by underlining the potential of services in the cleaning-up of the economy. Stahel (1982, 1997, 2010) develops the argument that "servicing" minimises the use of new inputs, and maximises the use of a product over its life-time, while benefiting both manufacturers (who retain control over assets, enhancing their maintenance and recovery) and consumers (who pay only for benefits).

Additional contributions include several concepts following the 3R principles of "reduction, reuse and recycle", which also share the notion of closed loops, including: the "zero emission" concept, which refers to systems whereby everything has its use and natural cycles are emulated (Pauli, 1997, 2010); the further development of the "cradle to cradle" model by Braungart and McDonough (2002); and the "zero waste" concept, whereby waste is diverted from landfills (Curran and Williams, 2012; Zaman, 2015; ZWIA, 2009).

With roots in different ideas and schools of thought, the CE thus emerges today as a wide-ranging concept, and all these various contributions must be considered in their specific contexts, as the CE has "different meanings and different roles and responsibilities for different stakeholders" (EIO, 2016, p. 9).

2.4. The "Circular Economy"

As the CE concept entered the policy arena it received a new boost. Germany showed an early interest in CE initiatives; for instance, its "Closed Substance Cycle and Waste Management Act" of 1996, tried to ensure environmentally-friendly schemes of waste disposal. In Japan the Basic Law for Establishing the Recycling-based Society of 2000 created a legal framework to induce a more recycling-based society (Preston, 2012; Su et al., 2013, p. 216). It was also made more practically relevant when it started to be discussed in China in 1998, and afterwards when it formally entered the language of the central government in 2002, as the country became the first to enact explicit policy regarding the CE (Geng et al., 2009b; Mathews and Tan, 2011; Zhu et al., 2010). Between 2005 and 2007, the CE concept was fostered through "two batches of circular economy pilots", in order "to promote circular economy philosophy into action, including key industries, key

Table 2		
Examples	of CE-related	concepts.

CE related concepts	Links with CE	Focus	References
Closed-loop economy	"() man must find his place in a cyclical ecological system ". "() highlighted the potential of a closed-loop economy impact on competitiveness, job creation, resource savings and waste prevention".	Focus on the need to "close" the loop in economical systems	Boulding, 1966, p. 9 Stahel and Reday-Mulvey, 1982, p. 93
Industrial ecology	"By analogy with natural ecosystems, an industrial ecology system () maximizes the economical use of waste materials and of products at the ends of their lives as inputs to other processes and industries".	Focus on emulating natural processes "closing the loop" in industrial systems	Frosch, 1992, p. 800
	"Industrial ecology involves designing industrial infrastructures as if they were a series of interlocking ecosystems".		Tibbs, 1993, p. 3
	"Moving from linear throughput to closed-loop material and energy use are key themes in industrial ecology".		Ehrenfeld and Gertler, 1997, p. 68
Symbiosis	"() industrial symbiosis (IS) can be categorized as a concept of collective resource optimization based on by-product exchanges and utility sharing among different colocated facilities"	Focus on industrial clusters and synergies	Jacobsen, 2006, p. 240
	Industrial symbiosis engages "() traditionally separate industries in a collective approach to competitive advantage involving physical exchange of materials, energy, water, and by-products".		Chertow, 2007, p. 314
Natural Capitalism	"Natural capitalism recognizes the critical interdependence between the production and use of human made capital and the maintenance and supply of natural capital ".	Focus on environmental and economic benefits of more effective manufacturing processes, reuse and recycle of materials	Lovins et al. 1999, p. 3
Cradle to Cradle	"If humans are truly going to prosper, we will have to learn to imitate nature's highly effective cradle-to-cradle system () in which the very concept of waste does not exist ".	Focus on design, since the conception stage, of competitive services/ products without negative environment impact	Braungart and McDonough, 2002, p. 103
Zero Waste	"Zero Waste means designing and managing products and processes to systematically avoid and eliminate the volume and toxicity of waste and materials concerne and recover all resources and not hum or hum them"	Focus on limiting waste and diverting it from landfills	ZWIA, 2009 unpaginated
	"Zero waste is a unifying concept for a range of measures aimed at eliminating waste and challenging old ways of thinking. Aiming for zero waste will mean viewing waste as a potential resource with value to be realised. rather than as a problem to be dealt with ".		Curran and Williams, 2012, p. 3
	"At this moment, ZW strategy is targeted toward zero landfills through diverting waste from landfills".		Zaman, 2015, p. 17
Functional Service Economy	"A functional economy () is one that optimizes the use (or function) of goods and services and thus the management of existing wealth (goods, knowledge, and nature). The economic objective of the functional economy is to create the highest possible use value for the longest possible time while consuming as few material resources and energy as possible".	Focus on new business models	Stahel, 1997, p. 91
	"The Functional Service Economy is a set of innovative business models that integrate products and services () to create health and jobs with considerably less resource consumption ".		Stahel, 2010, p. 2

Note: Main linkages with the CE concept highlighted in bold.

areas, key enterprises and urban demonstrations" (Dong et al., 2013a, p. 228). In 2008, the Circular Economy Promotion Law was approved, coming into effect in 2009, to improve "resource utilisation efficiency, protecting the natural environment and realising sustainable development" (Geng et al., 2012, p. 216). This orientation was reinforced in the 12th Five-Year Plan (2011–15), focusing on cleaner production and eco-industrial park development (Geng et al., 2009b; Shi et al., 2010; Xue et al., 2010).

By entering the western policy arena, for example through the 2015 "EU Action Plan for the Circular Economy" (EC, 2015), and by being sponsored by think-tanks (such as the World Economic Forum -WEF) and private institutions (such as the Ellen MacArthur Foundation -EMF), the CE became a topical theme undergoing intense study. This momentum seems to be related to several aspects. To start with, the CE concept has been considered more helpful and tangible than other approaches in solving environmental problems, offering at the same time both economic benefits and business solutions (Sauvé et al., 2016). Its potential for job creation, improved resource productivity, trade balance, and CO₂ emissions reduction, has been highlighted by policies in many countries, namely in Finland, France, the Netherlands, Spain and Sweden (Wijkman and Skånberg, 2015). Portugal also begun a public consultation process regarding its CE Action Plan for 2017–2020 (Grupo Interministerial Economia Circular, 2017). The concept has also been taken as an actual policy enacting device benefiting from several funding opportunities within the EU Circular Economy Action Plan (EC, 2017, 2015).

This does not mean that the CE is a consensual concept, or even that its definition is settled (J. Kirchherr et al., 2017). Many different recent definitions can be found, from international organisations, non-government organisations and academia (Table 3). Nonetheless, the definitions do highlight a set of core elements which characterise the CE as encompassing: i) input minimisation and efficient use of regenerative resources (material and energy efficiency as well as sourcing and prioritising the use of renewable and non-hazardous materials); ii) life cycle extension and systems reconceptualization (repair, re-conditioning and re-manufacturing options; procurement, new business models based for instance on sharing or re-use; design - from policy design to life-cycle approach and eco-design); iii) Output reduction valorisation and waste minimisation (recycling, networks of recovery, and valuing by-products and waste).

These components make up the CE, as a system deliberately designed to be restorative, replacing the end-of-life concept of the linear economy with new circular flows of reuse, restoration and renewability, in an integrated process, encompassing the entire value chain. In economic terms, the CE enables competitiveness through new ways of achieving more effective resource allocation, utilisation and productivity. Environmentally, the CE decreases negative externalities, and socially, it generates not only

Table 3

Examples of definitions of the CE.

Some of the most recent examples of definitions and descriptions of the CE		References
Circular economy	Regarding Chinese implementation of CE, it is defined as "() the realisation of a closed loop of materials flow in the whole economic system".	Geng and Doberstein, 2008, p. 232
	In China "The term 'circular economy' () is a generic term for reducing, reusing and recycling	Standing Committee of the
	activities conducted in the process of production, circulation and consumption".	National People's Congress (China), 2009, Article 2
	"It incorporates myriad strategies to achieve greater efficiency through economies of systems integration ".	Geng et al., 2012, p. 216
	One of the most used CE definitions is that of a "() system that is restorative or regenerative by intention and design ".	EMF, 2014, p. 12
	In Europe, CE has been defined as a way to keep "() the added value in products for as long as possible and eliminate waste ".	EC, 2014, p. 2
	The concept has integrated policy discourse as a way to "() boost the EU's competitiveness by protecting businesses against scarcity of resources and volatile prices, helping to create new business	EC, 2015, p. 2
	opportunities and innovative, more encient ways or producing and consuming . Regarding CE characteristics "() essential elements of a circular economy () include: refurbish ,	EIO, 2016, p. 10
	and minimisation of hazardous substances) plays also a prominent role"	
	"Central elements of the circular economy include remanufacturing and product life-cycle extension	UNEP, 2016, p. 246
	schemes such as re-use and refurbishment".	, , , , , , , , , , , , , , , , , , ,
	"() the concept of a circular economy (CE) is considered as a solution for harmonizing ambitions for	Lieder and Rashid, 2016, p. 37
	economic growth and environmental protection".	
	"By promoting the adoption of closing-the-loop production patterns within an economic system CE	Ghisellini et al., 2016, p. 11
	achieve a better balance and harmony between economy, environment and society".	
	CE "as a regenerative system in which resource input and waste, emission, and energy leakage are	Geissdoerfer et al., 2017, p. 759
	minimised by slowing, closing, and narrowing material and energy loops. This can be achieved through	
	long-lasting design, maintenance, repair, reuse, remanufacturing, refurbishing, and recycling".	
	"A circular economy describes an economic system that is based on business models which replace the	Kirchherr et al., 2017 pp. 224-
	'end-of-life' concept with reducing, alternatively reusing, recycling and recovering materials in	225
	production/distribution and consumption processes, thus operating at the micro level (products,	
	companies, consumers), meso level (eco-industrial parks) and macro level (city, region, nation and beyond) with the sim to accomplish sustainable development, which implies spectra environment-	
	Devolut, with the aim to accomption sustainable development , which implies creating environmental quality, economic prosperity and social equity, to the benefit of current and future generations."	
	quality, economic prosperity and social equity, to the benefit of current and future generations.	

Note - Main CE characteristics highlighted in bold.

employment opportunities, but also, new "consumer" concepts (EMF, 2012, 2013).

In spite of its broad scope, arriving at a clear and compact definition of the CE remains somewhat elusive. A working definition of CE, in tune with the reviewed strands of analysis, could be given in terms of it being an approach towards sustainable development. This approach is achieved through several strategies aiming to reorganise production and social systems into regenerative environmentally-sound closed circuits. Its main characteristics are focused both on resource and waste minimisation, as well as processes of production and consumption designed from the outset for efficiency, reuse, repair, and recycling.

Building on the contribution of Freeman and Louçã (2001, p. 124), we take CE as a form of *clean congruence*, i.e. a state of compatibility between technological and socio-institutional subsystems that overcome the unresolved mismatches of a Fordist, carbon-intensive, depletion-prone era. As a particular type of positive congruence, CE provides the most favourable, enduring and self-reinforcing conditions for sustainability.

Three levels of analysis have been presented in the literature, on the basis of which the depth or granularity of CE implementation can be appreciated (Ghisellini et al., 2016). At a *micro* level, the CE focuses on individual actors, particularly companies (Yuan et al., 2006; Zhu et al., 2010). Examples include: eco-design and cleaner production strategies; resource efficiency initiatives; labelling systems, and; sustainable production and consumption methods (Geng et al., 2009b, 2012). At the *meso* level, the focus is on actor interaction especially inter-firm networks: industrial symbiosis; eco-industrial parks; green supply-chain management and reverse logistics (Zhu et al., 2010). As for the *macro* level, the CE is theorised at a national or global scale, with an emphasis on legislation; regulatory impact analysis; zero waste regimes; and recyclingoriented societies (Ghisellini et al., 2016; Zhijun and Nailing, 2007).

CE is therefore here considered as: a) an integrative concept for attaining "clean congruence" by guiding new institutional set-ups that match environmental considerations with socio-economic performance while promoting techno-economic development that is not depending on the consumption of finite resources; b) a multi-level framework (micro, meso and macro) that re-shapes and re-directs production and business models toward resilience and sustainability; c) an encompassing notion calling for specific actions towards the minimisation of resource extraction, maximisation of reuse, increased efficiency, enhanced waste recycling and the development of new business models.

2.5. Eco-innovation and the circular economy: linking the concepts

If cheap resources for widening markets supported the 20th century's economic growth, the first decades of the 21st century brought rising price volatility and geo-economic uncertainty (Dobbs et al., 2011). Meanwhile, even if recycling is now seen as indispensable, waste production remains largely unchecked (WWF, 2014). Palliatives may not be enough, as global consumption has been increasing dramatically in the last two centuries and is expected to triple by 2050 (Vanner et al., 2014). New global trends are emerging, such as tighter environmental standards and consumer sensitivity to climate change. In this context, the concept of a new economic model, working in closed-loops, encouraging and encouraged by innovation throughout the whole value chain is advocated as an alternative solution for minimising waste of materials and energy in a world that remains competitive and dynamic, but finite (Potočnik, 2014; UNEP, 2006, 2011).



Fig. 1. Relationships between EI and the CE.

The EU, since the adoption of the Lisbon Strategy in 2000, has been actively involved in the development and implementation of a "greener" sustainable economy and society, assuming a global leadership role in this regard. Its most recent efforts concerning the promotion of a transition to sustainability have focused on a number of flagship projects and action plans concerning EI (EC, 2011a; EIO, 2011, 2013), resource efficiency (EC, 2011c; 2014) and, most recently, the CE (EC, 2015). The pursuit of a CE is now central within the EU agenda, with the Commission's Circular Economy Action Plan stressing the EU's commitment and support for CE, but also recognising the close connection with innovation, and especially EI (EC, 2017). It is thus now argued that the CE is contingent "(...) on adopting a systemic approach to eco-innovation that encompasses value and supply chains in their entirety and engages all actors involved in such chains" (EC, 2016, p. 73).

Transforming production routines and consumption habits through an endless rewiring of loose ends of various activities is a dynamic enterprise (EMF, 2013, 2012). El is identified as a key way for doing so, through the development of new products and processes based on new technologies, as well as new business models, based on novel organisational forms and marketing schemes (Tregner-Mlinaric and Repo, 2014). El-CE connections, the key focus of this study, can thus be explored (see Fig. 1).

The connections between these two concepts are, nevertheless, complex. Both still encompass several related terms and have somewhat vague boundaries. Even if one can intuitively argue that EI and CE are closely related, and assume that achieving CE without EI is unlikely, it remains to be seen in what ways this is so. Not all EI is related to the CE, and EI might also have different impacts in several areas of the CE. While their relation is undeniable, a deeper analysis could make use of the specific aspects already pointed out, namely EI *targets, mechanisms* and *impacts* (horizontal axis) and the *micro, meso* and *macro* levels of CE (vertical axis). Fig. 2 brings these categories together so as to explore the *dimensions* of EI (horizontal axis) for each *level* of the CE (vertical axis). The ensuing literature review provides supportive evidence, whilst distilling the practical insights. A deeper understanding of the overlap between EI and the CE may help to articulate how a closed-loop, production-utilisation

congruence requires thorough implementation of specific types of change. These self-reinforcing patterns (that can be understood as forms of "clean congruence") can be attained at distinct levels (macro-meso-micro), which may be mapped and monitored as policy, and other decision makers, seek strategies for transition toward a CE. The generic term of "clean congruence" refers to the process of dealing with mismatches at a variety of levels between ecological and economic sustainability in the context of an emerging (green, innovative) techno-economic paradigm.

3. Methodological approach

This paper assumes that it is an appropriate time for investigating the overlap between different streams of research that converge on transition-driven innovation. A literature review, following and adapting several prior methodological contributions (Bocken et al., 2014; Boons et al., 2011; de Jesus and Mendonça, 2018; Khan et al., 2003; Patala et al., 2014), can help the development of concepts linking the converging clusters of ideas broadly understood as EI and the CE.

3.1. Data criteria and collection

Data was obtained from scholarly peer-reviewed journals. A circumscribed corpus was identified by using two of the most widely-used databases of academic journals, namely the Web of Science (WoS) Core Collection¹ and Scopus.² Material was

¹ The WoS TM Core Collection is contained within the Web of Knowledge database platform of bibliographic references, produced by the Institute for Scientific Information (ISI), covering over 12,000 of the highest impact journals worldwide in the fields of sciences, social sciences, arts, and humanities, including Open Access journals and over 150,000 conference proceedings, being one of the most generally acknowledged sources of data for bibliometric studies (Franceschet, 2009; Moya-Anegón et al., 2007).

² Scopus is the largest abstract and citation database indexing the greatest number of peer-reviewed journals (Falagas et al., 2008), around 21,500, from more than 5000 international publishers (Elsevier, 2014), having a more European focus (Chappin and Ligtvoet, 2014).



Fig. 2. Levels of the CE and dimensions of EI. Source - Elaborated from several contributions (Geng et al., 2012; Geng and Doberstein, 2008; OECD, 2010; OECD/Eurostat, 2005)

identified through a keyword Boolean search on title, abstract, and keywords of articles and reviews, written in English, in the assumption that this would identify all the latest relevant global research. The goal was to ensure that relevant peer-reviewed publications were found. As the interconnections between EI and the CE are still not clearly defined, it was acknowledged, from the onset, that the use of the target terms alone carried a risk of exclusion and bias. As section 2 stated, there are several concepts closely connected to EI and the CE. Which ones should then be chosen as keywords, and which ones excluded?

as

In an initial exploratory exercise, in order to minimize the biases of subjectively choosing a closely connected concept over another, the first query searched only: "*innovat*" AND ("circular economy" OR "circular-economy" OR "circul* economy*").³ This search identified a total of 21 downloadable articles across the two databases (several articles appeared in both databases and were only counted once). Although this effort produced a very limited set of results, the articles enabled the identification of 86 unique keywords (provided by the authors of the articles). As keywords contain critical and concise information regarding the substance of each article, they were used as guidance for finding other relevant works. By analysing keywords used at least twice, the descriptors "industrial symbiosis", "industrial ecology", "urban symbiosis" and "eco-industrial park" were highlighted. These new descriptors were then used, in association with "*innov*", resulting in the retrieval of an additional set of documents (120 new articles). A grand total of 141 articles, published between 1992 and 2015, were thus assembled as the final corpus for analysis. Fig. 3 illustrates the path followed.

3.2. Analysing the articles

Drawing on the propositions underpinning the EI and CE bodies of research (section 2), the articles were analysed and read in full aiming to identify EI dimensions (i.e. target, mechanisms and impact) per levels of the CE (i.e. micro, meso, macro). These

 $^{^{3}}$ The search was carried out on 27 March 2015, and then updated on 13 $\,$ November 2015. For a further exploration of a part of the corpus, concentrating explicitly on drivers and barriers of pro-CE eco-innovation, see Jesus and Mendonça, 2018. The Boolean operator * is used to enable the return of expressions that begin with the word truncated by the asterisk.



Fig. 3. Layout of the research design.

categories (summarised in Fig. 2), are used to organise the extraction of meaning and trends from the 141 papers.

Categorisations are not straightforward, and their application requires judgement. Regarding EI impacts, for instance, the available literature stresses the difficulty of outlining and measuring it accurately (OECD, 2010). As such, inspired by examples identified in the corpus itself, some EI impacts are typified (Table 4). Here too it was necessary to make choices. Although the impacts are normally divided in the literature by their direct or indirect effects, the types identified could often have both effects. For instance, increased reuse and/or recycling of materials has a direct impact, in terms of reducing pollution and waste production, but may also have an indirect impact, in terms of improving awareness and implementation of CE models.

As for the CE, several of the articles could fall under more than one level of "circularity". For example, cleaner production at a company – the micro level –, when addressed from a government perspective (i.e. fostering the implementation of those initiatives). could be seen as macro level (Geng et al., 2010b). Similarly, as regards eco-towns and urban symbiosis, although other articles usually place these at the macro level (Ghisellini et al., 2016), it was here judged to be better to identify these as meso level, emphasising the cooperation between the city and other actors. As such, choices had to be made when allocating articles to a unique category. The overall goal of the analysis, however, is not to gather exhaustive examples, or to carry out a definitive analysis, but rather to convey the core thinking behind the demonstrative cases, and highlight the major patterns that can be gleaned from the literature, so as to enable the emergence of new conclusions regarding the poorly understood connection between EI and the CE. The inescapable degree of subjectivity involved in implementing the survey criteria was not, therefore, viewed as overly problematic.

4. Eco-innovating towards a circular economy: results of the literature review

In an aggregated overview of the articles it was found that most were published between 2006 and 2015 (83%),⁴ which shows that there is a growing interest in these fields (Fig. 4). Using the operational definitions, and the proposed analytical framework, the corpus was examined by CE level, focusing on the role of EI. The macro/meso/micro organising principle for unpacking EI-CE connections allows us to give structure to the findings. The aim was to identify developments in the literature, as well as research gaps and policy prospects.

4.1. CE at the macro level

Examining the sub-set of articles categorised at the macro level, four important considerations stand out: 1) the CE emerges as a multidisciplinary, difficult to define, concept; 2) governance and public policies have a central role in supporting and promoting EI and the CE; 3) at the wider national and transnational scales, resource efficiency and waste management are particular concerns; 4) EI appears to be an enabler of the transition to a CE.

CE is indeed characterised as a wide-ranging concept, still rather difficult to define: "circular economy does not have a single

 $^{^{\}rm 4}$ As the search was updated the 10th September 2015 this can justify the apparent drop in 2015.

Table 4	
Key characteristics of EI impacts.	

Direct and indirect impacts	Description of impacts	
Resource productivity (materials/ energy) Transition to renewables and low impact materials and energy Increase material reuse, and/or recycling Waste/ pollution/ emissions reduction Reduce costs and or generate revenue Improve awareness and implementation of circular models	Material and energy efficiency New types of energy sources and materials Durability and valorisation Lessening of pollution, waste and emissions generation Expenditure reduction or profit generation Circular models accentance	
Indicator developments, policies' applications	Development of indicators and policies	

Source: Elaborations on the corpus.

definition, it generally stresses closed flows of materials, and increased efficiency in the use of raw materials and energy" (Matus et al., 2012, p. 194). Contributions from several different schools of thought add to its intellectual development, from industrial ecology, systems theory, global environment studies, environmental innovation, spatial planning, societal transitions, ecological modernisation, technology policy, and innovation management.⁵ The links of CE to a diverse economics background are also evident, being associated with fields such as evolutionary economics and ecological economics – i.e. heterodox research programmes (del Río et al., 2010; Koenig and Cantlon, 1999; van den Bergh, 2013), as well as environmental economics – i.e. more mainstream approaches to the environmental agenda (Su et al., 2013).

At this level another issue raised was the important role of governments in: providing context; ensuring coordination, and; leading the way in the promotion of new industrialisation models that are more efficient, less polluting and involve less exploitation of resources. Progress in science, technology and innovation is identified as a way for developing countries to advance their overall catch-up process (since they have the potential to leapfrog, at least in the environmental-economic nexus), and also a way for developed countries to increase well-being and reduce vulnerability to resource price shocks (Cheng, 2007; Geng et al., 2009b, 2012). Governmental action is, therefore, considered fundamental in managing "different initiatives, enacting appropriate regulations, stipulating feasible guidelines and standards, providing substantial financial support and carrying out international collaboration" (Geng et al., 2010b, p. 1507). Governmental action emerges as both an instrumental driver, in framing pro-CE behaviour and transitionfriendly networking capabilities, as well as a barrier, when failing to "enable" a CE context.⁶ Since a CE remains a concept under construction, misunderstandings and misaligned policies are possible. Regulatory frameworks (i.e. taxes and incentives) must provide clear objectives in terms of environmental performance, helping to address market failures and allowing CE initiatives to prosper. At the same time, public agencies play a crucial role in ensuring planning and institutional guidance (for example, infrastructure provision and a conducive legal system), as well as by providing R&D support, enabling information exchange, encouraging the engagement of actors and promoting awareness, e.g. amongst enterprises, universities and wider society (Cheng, 2007; Nguyen and Ye, 2015).

In the promotion of a CE, several countries have already acted at a policy level, promoting legislation with CE effects, for instance the



Fig. 4. Number of academic published papers per year.

Source - Elaborations on the corpus; applies to all tables and figures from now on, unless otherwise stated. **Note** - N = 141.

EU action plan for a CE (EC, 2015). Asian countries have also demonstrated an awareness of the CE agenda, particularly Japan and China. The latter was a pioneer of explicit legislation regarding CE (Dong et al., 2013b), making it a key national policy and a regulatory priority, in particular as a vector for focusing on cleaner production and eco-industrial parks development (the *12th Five-Year Plan*, 2011–15).

Emerging policy avenues have underlined the need to move away from the existent resource-based paradigm. As consumption has risen, in both developed and developing countries, recycling and reuse has been identified as vital in closing the loop (Graedel and Cao, 2010). This transformation is considered dependent on innovative technologies, as well as new organisational forms, to manage resources and waste (Geng et al., 2014; Giannetti et al., 2004; Zhijun and Nailing, 2007). Awareness of the intrinsic value of waste and "the extent of knowledge that (...) led to technological innovation for reuse" (Park and Chertow, 2014, p. 47) has become essential. If throughout human history waste has been recognised in a negative sense (associated with unwanted, unusable, worthless materials, lacking economic value or potential), current challenges reinforce the need for a rethink. In a CE waste is meant to be minimised (Köhler et al., 2011; Levänen, 2015) and rather returned as an available resource⁷ in a process that is efficiently macromanaged in order to guarantee further community development. The need for novel management practices in the production process is stressed, i.e. production needs to be organised more broadly, so that it transcends the linear input-to-output sequence (Jones et al., 2013).

Technology-driven El is considered as an enabler of new ways of reusing and recycling substances, giving them other industrial applications (Wen et al., 2007). However, the CE is more than just about re-engineering existing processes (i.e. incremental change of existing components); importantly, it is also about re-wiring (i.e.

⁵ To enable easier reading, when there are more than three references together these are gathered in footnotes, here: (Baas and Hjelm, 2015; Bakshi et al., 2015; Cohen, 2006; Deutz, 2009; Huber, 2000; Koenig and Cantlon, 1999; Körhönen et al., 2004; Körhönen, 2008; van den Bergh, 2013).

⁶ (Andrews and DeVault, 2009; Bergquist et al., 2013; Cheng, 2007; Heyes and Kapur, 2011; Yarime, 2007).

⁷ (Birat, 2015; Corder et al., 2015; Fichter and Hintemann, 2014; Muñoz et al., 2008).



Fig. 5. El Target, Mechanism and Impact at the CE Macro level, analysis of the corpus. Note - N = 80.

changing the architecture of) the whole system of supply and demand. More cost-effective, less environmentally-harmful innovations hinge upon the creation of realistic market opportunities (Brils et al., 2014), as well as the design of new processes and products,⁸ while information and communication technologies are crucial in product/service "dematerialisation", product tractability and performance monitoring (Erdmann and Hilty, 2010; Maurizio Catulli, 2012; Moreno et al., 2011). Given its role in decreasing the environmental impact of economic activities, EI is understood as a major new avenue for introducing systemic novelty in the transition towards more sustainable and viable counties, within an integrated vision of society, economy and environment.⁹ Overall, process and organisational innovations are the types of EI more emphasised at the macro level (Fig. 5). Technological EI emerges as critical, mostly in the form of incremental mechanisms based on the redesign of existing products and production methods, focusing particularly on increasing resource productivity (see Fig. 5).

Macro-level "circular" EI, however, is also characterised by mixed environmental results (Vivanco et al., 2014). There are rebound effects. For instance, low carbon technologies use rare materials, such as lithium. The availability of these materials can become an environmental and procurement problem for nations and regions. With regard to the UK electric vehicles market, Busch et al. (2014) provide an example that material flows should be holistically managed so as to avoid constraining the long-term potential for improving the reuse, re-manufacture and recycling of the materials involved. EI is also hampered by: high initial costs (Busch et al., 2014; Mirabella et al., 2014; Reh, 2013); limited public and business awareness (Heiskanen and Lovio, 2010; Jones et al., 2013; Riding et al., 2015), and potentially; regulatory mismatches and conflicting interests between economic and environmental agendas.¹⁰

Similarly, and in spite of playing an important role, science and technology per se (i.e. El disconnected from the broader context) are not considered sufficient conditions for the transition to new paradigms. Whereas technological innovation is believed essential for boosting resource efficiency, as well as production and waste minimisation, non-technological innovation is still deemed essential for "selling" new products and services (Dewick et al., 2007). A systematic approach to change, addressing the societal and contextual settings, is thus highlighted as crucial (Huesemann and Huesemann, 2007). More than revolutionising the existing economic structure, the EI mechanisms stressed at the macro level focus on evolutionary changes towards a "clean congruence", based on incremental redesign and modification of existing systems spanning different sectors and value chains. This seems compatible with the development phase of a CE when several products, industries and business models are emerging (Blowfield and Johnson, 2013). Within this transition, the emergence of new technologies is complementary to wider social, economic and legal/political developments, including increased public awareness, new regulations, and changes in market supply and demand. At a macro level, this phase can be characterised as a temporary period of reorganisation and reconfiguration of a country's techno-economic systems, involving all societal actors.

4.2. CE at the meso level

The meso level addresses networks and interactions. Moreover, CE's own nature, as an integrative multi-actor approach, points to the importance of networks for: building capacity; increasing cooperation in research and investment; sharing materials and by-products, and; managing common utilities and infrastructures. The establishment of these networks is generally motivated by agents interested in cost reduction, economies of scale, and lesser exposure to resource price volatility, and is a determining factor in the implementation of a truly CE. At a meso level, the CE links with several concepts related to the establishment of cooperation and alliances, from which the corpus emphasises those in or within: industry (e.g. industrial symbiosis and eco-industrial parks); value chains (e.g. sustainable, environmental and "green" supply chains,

⁸ (Anastas and Lankey, 2000; Barberio et al., 2010; Fiksel, 2002; Grundmann et al., 2013; Jin et al., 2004; Matus et al., 2012; Ogunseitan, 2007; Reh, 2013; Thomas et al., 2003; Wen et al., 2007).

⁹ (Carrillo-Hermosilla et al., 2010; Cheng, 2007; del Río et al., 2010; Ganapathy et al., 2014; Tombesi, 2006).

¹⁰ (Busch et al., 2014; Jones et al., 2013; Matus et al., 2012; Riding et al., 2015).

and extended producer responsibility); local-government initiatives (e.g. eco-towns and urban symbiosis). The emphasis given to these concepts, rather than others, may be related to the scope of the corpus itself, covering largely European and Asian examples.

Industrial symbiosis is based on an "industrial systems integration" approach (Geng et al., 2014). It focuses on the potential of networks for exchanging materials and by-products, as well as for sharing management of common utilities and infrastructure for water, energy and waste, between several actors (van Berkel et al., 2009). The sharing of services, such as transportation and infrastructure, and the brokering of by-products (so that the waste from one industry waste becomes the input of another), results in pollution mitigation, decreased use of materials and energy, and cost reductions, and thus creates both economic and environmental benefits. Kalundborg in Denmark is considered the pioneer model and inspiration, but there are already several other examples of industrial symbiosis.¹¹ Linked with industrial symbiosis, the notion of an eco-industrial park is also important. Eco-industrial parks retain the positive externalities of industrial parks, which arise from: businesses being located close together; economies of scale; inter-firm communication; centralised transportation, and; waste disposal infrastructure. However, they also add the potential for symbiosis regarding ecological considerations, related to minimising negative impacts in local resource depletion and pollution. In spite of geographical peculiarities (given that the definition and implementation of industrial symbiosis and eco-industrial parks can differ from country to country) (Boons et al., 2011) and varying stages of development (Chertow and Ehrenfeld, 2012), ecoindustrial parks have been found to foster symbiotic networks of cooperation between enterprises (Yu et al., 2015), thus actively promoting the CE at an industrial level (Zhu et al., 2015). For instance, at TEDA – Tianjin Economic-Technological Development Area (China) – the integration of the regional water cycle system provided recycled water to all of the area's users, demonstrating the potential of symbiotic relationships within the eco-industrial park (Yu et al., 2014). El is considered essential in the development of eco-industrial parks and industrial symbiosis, whilst, at the same time, these concepts have a role in the development of institutional, technological and business model innovations (Shi and Yu, 2014).

As for "sustainable supply chain management" (Gupta and Palsule-Desai, 2011; Ji et al., 2014; Zhu et al., 2010), "green supply chain" management (Mirhedayatian et al., 2014; Park et al., 2010), "closed-loop supply chain" management (Guide and van Wassenhove, 2009), extended supply chain (Zhu and Geng, 2013), and "extended producer responsibility" (Chen and Sheu, 2013; Lai et al., 2014), these concepts all focus on improving a product lifecycle via its supply chain. The objective is to reduce costs by sustainably managing the life-cycle of products from conception (e.g. less use of materials and energy in production, packaging and delivery) to end-of-life (e.g. reuse, reduced waste, creating recovery networks etc.). This involves the responsible incorporation of environmental considerations into supply chains from the outset, and the promotion of cooperation between companies, suppliers and customers, to "close the loop". The conversion of existing supply chains is supported by a set of technological developments that enhance resource efficiency, reuse and recycle, as well as organisational innovation leading to new distribution, collection and business models (Rashid et al., 2013).

Considering that urban growth is accelerating, especially in developing countries, and that cities play a role as both industrial and population centres, an integrated approach to cities is necessary for designing new ways of tackling environmental problems and mitigating pollution (Dienst et al., 2013). In China, in particular, the problems of the so-called "resource-based cities" (i.e. cities primarily orientated towards extractive and/or resource-intensive industries) have drawn attention given the importance of integrative strategies for moving towards more "circular" cities (Dong et al., 2013a). In this regard, the concepts of urban symbiosis and eco-towns have extended the network rationale to urban actors. This is an integrated view of urban infrastructure, maximising benefits arising from the interrelation between the city and its industrial context, as well as the possibilities within the city itself for fully capitalising on reinforcing infrastructural use, rather than duplicating resources.¹² In this domain, an innovation-friendly environment, as well as efforts from several actors (including governments and companies), are important factors in the development of low-carbon cities (Dong et al., 2013a). Innovation in these processes involves enhancing the ability to develop both "software projects" (e.g. town planning, community recycling, and outreach activities) and "hardware projects" (e.g. innovative recycling facilities and associated infrastructure) (Chen et al., 2012).

Closing the loop at the meso level therefore seems dominated by inter- and cross-sectoral pooling of infrastructural resources (i.e. eliminating wastage in overhead capital), as well as by the maximisation of synergies across different value chains (i.e. interactions between production systems and agents). Promoting cooperation and interrelations between geographically-close actors, companies and organisations, is considered an effective way of achieving a more circular system, with better use/reuse of resources. It seems interesting to note that, in regard to these relationships, the role of business associations is never mentioned in the corpus, while one could expect these and other entities (such as consumer organisations) to play a role in bringing companies together, so as to increase collective environmental efficiency.¹³

Overall, El at the meso level is described as a facilitator of sectorial or regional systemic integration, enabling new ways of sharing services, utilities and by-products, which in turn provides new ways of promoting cooperation.¹⁴ Green and transformative innovation is key for engaging in financial engineering (i.e. responding to high initial costs and capital investments), as well as for identifying symbiotic links between organisations and sectors (i.e. synergies), and for addressing technical issues such as solid waste, air pollution, water contamination and noise pollution (i.e. bottlenecks for transition).¹⁵

At this level the literature points to the importance of this green and transformative innovation in attaining a "clean congruence" mostly based on the organisational dimension, on incremental mechanisms (redesign of organisations and processes is particularly stressed in the articles) and on targeting resource efficiency, material reuse and recycling (Fig. 6). The transition toward CE seems therefore plough on "green collective innovation"

¹¹ (Chertow and Ehrenfeld, 2012; Geng et al., 2009a; Liu et al., 2012; Mathews and Tan, 2011; Park et al., 2008; Patnaik and Poyyamoli, 2015; Shi et al., 2010; Shi and Yu, 2014; Yu et al., 2014; Zhu et al., 2015).

¹² (Chen et al., 2012; Dong et al., 2013a, 2014; Geng et al., 2010a; Niza et al., 2009; van Berkel et al., 2009).

¹³ The general role of such intermediate associations in the innovation process has been emphasised in a small amount of relevant literature (see, e.g., Dalziel, 2006; and Watkins et al., 2015).

¹⁴ (Baas, 2011; Bristow and Wells, 2005; Desrochers, 2004; Dong et al., 2014; Gupta and Palsule-Desai, 2011; Killerby et al., 2007; Lombardi and Laybourn, 2012, 2012; Mirata and Emtairah, 2005; Paquin and Howard-Grenville, 2012; Park et al., 2008; Ruiz Puente et al., 2015; Short et al., 2014; Simboli et al., 2014; Watkins et al., 2013; Yu et al., 2015; Zhu et al., 2010).

¹⁵ (Cecelja et al., 2015; Geng et al., 2009a; Hewes and Lyons, 2008; Liu et al., 2012; Mathews and Tan, 2011; Park et al., 2008, 2008; Patnaik and Poyyamoli, 2015; Raafat et al., 2012, 2013; Shi et al., 2010; Shi and Yu, 2014; Sterr and Ott, 2004; van Berkel et al., 2009; Yu et al., 2014; Zhu et al., 2015).



Fig. 6. El Target, Mechanism and Impact at the CE Meso level, analysis of the corpus. Note: n = 43.



Fig. 7. El Target, Mechanism and impact at the CE Micro level, analysis of the corpus. Note - N = 18.

trajectories as a way to achieve "clean congruence" at the meso level.

4.3. CE at the micro level

The micro level focuses on specific agents' capabilities and involvement in CE. It comprises, nevertheless, the smallest part of the corpus, which was a surprise. Although this result could indicate an area of research that is still on its way to maturity, it may also be a sign of a methodological shortcoming. First, it may be that the most appropriate keywords for the micro level were not used; second, and more important, it may be that innovation at this level is very specific and unlikely to be published within scientific articles (patent data could be more revealing in this respect, for instance, something interesting but not pursued in Silva and Mendonça in UN, 2015). Within the corpus, this body of work focuses particularly on cleaner production, eco-efficiency, eco-design and new business models.

Cleaner production emphasises the application of processes, technologies and practices for minimising resource and energy consumption, as well as pollution, in order to accomplish a better overall efficiency within the organisation (Geng et al., 2010b). It includes green design as well as the introduction of clean energy and waste management technologies (Basu and van Zyl, 2006). Other practices such as eco-efficiency (i.e. production of goods or services with fewer resources and waste), and eco-design (i.e. design for the environment), similarly aim to design products with environmental considerations throughout their whole life-cycle, thus ensuring energy savings and pollution reduction.¹⁶ The literature cites practical examples ranging from the conservation of

¹⁶ (Collado-Ruiz and Ostad-Ahmad-Ghorabi, 2013; Matos and Hall, 2007; Mattinen et al., 2015; Mont, 2008; Sanyé-Mengual et al., 2014).

resources (Silva et al., 2015), to product design focusing on life-cycle aspects regarding materials usage, processing and maintenance, as well as communication with end-users (Sanyé-Mengual et al., 2014).

New business models based on leasing, rental and "sharing" services, also emerge in this literature, focusing on the replacement of capital ownership and proprietary models. In areas as diverse as housing, transportation and communication, these business models promise more efficient use of resources, extended lifespan of products, and greater reuse of materials at the end-of-life of products (Albu, 2011; Short et al., 2014).

At the micro level, the El of goods and services is particularly stressed, especially in an incremental mode, in terms of both increasing resource efficiency (Adams and Ghaly, 2007) and ecodesign (Cerón-Palma et al., 2012; Sanyé-Mengual et al., 2014). Nonetheless, radical alterations are also believed to be necessary as the transition to new sustainable ways of living implies the genuine transformation of the *status quo* (Fig. 7).

Technological EI mechanisms are established as tools for addressing bottlenecks in product durability and quality, in designing goods with longer usability spans, and addressing problems of decreasing efficiency over time (Adams and Ghaly, 2007; Mattinen et al., 2015). Designing optimal product life scenarios requires an in-depth knowledge of durability and the replacement schedule of parts. However, it also creates the possibility of constant upgrades (Bakker et al., 2014). Alternative, less expensive ways of reusing and re-manufacturing products are indispensable, since the costs of re-manufacturing are still often higher than the costs of production using virgin materials. Likewise, new ways of limiting the extra (economic and environmental) transportation costs involved in product reuse and remanufacturing, are essential in order to make these activities more viable (Mont, 2008). But, even if the technology already exists, other kind of innovations may be needed to overcome several reasons that often prevent more sustainable designs. Consumers are still mostly unaware of the choices available (Finster et al., 2001). The lack of transparency and credibility coming from dissatisfaction with empty greenwashing rhetoric also hampers the development of "green markets" and the willingness of customers to pay for "green" goods and services (Lemke and Luzio, 2014).

Non-technological EI, promoting new organisational models may support new schemes for increasing product use intensity through sharing and pooling. At the same time, marketing innovations can enable new ways of distribution, usage and perception for products and services - e.g. monthly payments for the use of refrigerators, washing machines, concrete mixers, and other tools such as drills, saws, hammers etc. (Bakker et al., 2014; Ceschin, 2013; Mont, 2008). These softer type of innovation trends may create incentives for producers to: develop longer-lasting products; replace existing products with more efficient models; and even to upgrade already existing products when new technology becomes available (Mont, 2008). The role of marketing innovation is rather unexplored in the corpus. Similarly, the role of consumers as innovative agents is also not much addressed. This seems paradoxical, since consumers are an integral part of a CE, not only as demand-side actors, but also as an active part of global supply chains themselves.¹⁷ Hence, "dynamic CE business models" seem to be of the essence at the micro-level as a ways to operationalise "clean congruence" and enable transition.

4.4. Clearing the ground for "clean congruence": exploring the meaning and implication of EI-CE connections

The assembled corpus of articles enables some considerations about the main objectives of this paper: 1) to derive literaturebased working definitions and links between the EI and CE concepts; 2) to review and assess the relation between EI and CE's various levels, and; 3) to identify the types of EI which are most influential in driving the transition to a CE.

First, the definitional challenge is not a minor one. The CE concept suffers from vague boundaries and it includes inputs from several schools of thought (Matus et al., 2012). Several slightly different definitions, linking innovation to environmental questions, have also been proposed in the last few decades (Carrillo-Hermosilla et al., 2010). Although the latitude still present in the studies on EI and CE enables the integration of several strategies and perspectives under increasingly used labels, it may hinder their development and hollow out their meaning. Tentative definitions of CE and EI are thus required, which aid the development and use of both concepts; a need this paper addressed by proposing working definitions for both CE and EI.

As a broad framework, the relationship between CE and the notion of innovation is still not obvious. CE is an integrative multiactor approach in which EI (technological and non-technological based) is a tool in the transition towards a cleaner form of "congruence" bringing about a new "techno-economic paradigm" (in the sense of Freeman and Louçã, 2001). Some authors focus on CE strategies as drivers for EI: e.g. as a "leading principle for ecoinnovation, aiming at 'zero waste' society and economy" (Mirabella et al., 2014, p. 29). However, other authors highlight the causal role of EI within the CE; e.g. "the capacity of eco-innovations to provide new business opportunities and contribute to a transformation towards a sustainable society" (Carrillo-Hermosilla et al., 2010, p. 102).

The inherent relationship between these concepts appears in all the CE levels considered, which we take as pathways or trajectories towards achieving "clean congruence" (introduced here as a bridging concept between EI and CE literature). Moreover, the disaggregation by CE level enabled a more thorough and deeper analysis of the main features of EI. At a macro level, governance has a central role in providing context (Cheng, 2007; Geng et al., 2010b, 2012). By promoting EI-led CE policies related to waste management, infrastructure availability, science and technology improvements and public awareness, the government can be an inspiring actor. Governments may also have a coordination role in the movement towards a "clean congruence" at cross-sectoral and cross-regional levels; i.e. by avoiding wasteful lock-ins and mismatches that may lead to system failures and barriers to transition along broadly interdependent constituencies and value chains. At this level EI refers to the broad self-reinforcing combinations of socio-technological coalescing changes (i.e. "clean congruence") that allow transition to a CE to take place.

At the meso level, CE is considered to be contingent on systemic or transformative EI enabling new ways of "green collective innovation". That is, innovation that is based on multi-actor and multiexpertise comprehensive technological and non-technological (i.e. organisational and process) change. The latter involves creating new ways of sharing services, utilities and by-products, i.e. providing new ways to promote cooperation. This level highlights the importance of public policies and new ways of boosting cooperation between enterprises and public actors, promoting symbiotic links, addressing technical issues and overcoming institutional barriers (Boons et al., 2011; Cecelja et al., 2015; Raafat et al., 2013).

At a micro level, business strategies range from internal actions

¹⁷ The work on user-driven innovation is very relevant here (see, e.g. Von Hippel, 2005), and could have many applications in the area of EI and the CE.

of cleaner production (in energy and materials efficiency) to the development of new, more circular business models (i.e. servicebased user-producer relationships). At this level, the replacement of the "take-make-dispose" business model implies a greater emphasis on new products, servicing, resource pooling and marketing concepts with EI as a tool to address bottlenecks in product durability and quality, in designing efficient products and "dynamic CE business models". The sharing of business models and resource pooling schemes are emphasised, as they are especially dependent on pricing innovations and networking-empowered behaviours (Albu, 2011; Short et al., 2014). However, the role of users and citizens at large, so essential in a paradigm shift, is not yet really addressed in the corpus. This may point to the fact that, in spite of being essential for "circularity" efforts, these issues are seen as too narrow to be called "circular" per se. Research at the micro level is also constrained by the fact that the application of the notion of what a CE-based business concept can mean is still underdeveloped and in a state of flux.

Finally regarding the most influential types of EI driving the transition to a CE organisational and process EI seem generally well developed (especially at the macro and meso levels), whereas references to marketing innovations are scarce. Regarding EI mechanisms, even if authors do reiterate the need for more radical approaches, incremental EI is still predominant. Technological EI, in particular, is considered to be an enabler of change, and essential in the creation of a CE even if the transition is acknowledged to require more than just science and technology.

5. Key implications of the EI-CE connections

Our review points toward some key themes and main links between EI and CE, which helps to outline broad influential types of EI within specific levels of the CE. These intersections in turn allow a better understanding on how processes of innovation shape transition to a CE and inspire some preliminary considerations regarding policy and business implications. In particular, we relied on constructs such as EI-dimensions and CE-levels to outline how current research is pointing to pockets of "clean congruence", which in turn may provide guidance to policy (summarised in Table 5).

Regarding policy, the link between the CE and EI has been most explicitly addressed in recent years by the EU. Increased connections between the two concepts was apparent in the recent *EU Action Plan for the Circular Economy* (EC, 2015), following the prior resource efficiency agenda (EC, 2011c), as well as in the Eco-Innovation Action Plan (EC, 2011a). Moreover, policy measures related to regulatory and economic instruments are now closer to the policy implications identified in the corpus. These have been emerging in some EU countries, especially in aspects related to research, education and networking. Even if those efforts are not yet widely disseminated, the examples already in place constitute interesting opportunities for acquiring information about the practical application of pro-CE policy at the national and local levels, whilst also highlighting the differences between actors (EIO, 2016). Examples and characteristics, compiled in Table 5 are not

Table 5

Main features from the EI-CE literature, including "pro-CE" EI characteristics, types, and policy and business implications, by CE level.

	Key features from the EI-CE literature	El role	Major types of CE-inducing EI	Policy and business implications	Examples in the corpus
Macro	At a macro level, El refers to the broad self-reinforcing combinations of socio-technological coalescing changes (i.e. "clean congruence") that	El mechanisms focus on evolutionary mutations towards overall " <i>clean congruence</i> ".	Target/Type:- Organisational;	The importance of explicit public policies and new ways of streamlining cooperation between the public and private sectors.	Carrillo-Hermosilla et al., 2010; Cheng, 2007; del Río et al., 2010; Geng and Doberstein, 2008.
	allow transition to a CE to take place.	El's role as decreasing the environmental impact of economic activities, introducing all-round novelty in the transition towards more sustainable, and integrating vision of society, economy and environment.	Mechanism:- Technological.	Public agencies have a crucial role in planning, providing institutional standards and guidance (infrastructures provision/ conducive legal system). Pro-CE innovation policy is to provide R&D support, but also should facilitate peer-to-peer information exchange.	
Meso	At a meso level, innovativeness for circularity is a distributed and systemic process, where the potential for synergies within value chains and territories are ripe.	El as a facilitator of systemic integration, enabling new ways of "green collective innovation" such as sharing services and other schemes for maximising the value of common resources.	Target/Type:- Organisational;	Promoting the cooperation and interrelation of geographically close companies and organisations is considered to be an effective way of achieving a more circular system, with better use of energy, materials and resources.	Gupta and Palsule-Desai, 2011; Lombardi and Laybourn, 2012; Ruiz Puente et al., 2015; Watkins et al., 2013; Zhu et al., 2010.
		El is a way to re-direct and re-employ by-products among diverse industrial processes or actors	Mechanism:- Technological.	Strengthened cooperation between actors, and resulting synergies, limits exposure to resource price volatility, reducing costs and minimizing the use of non-recyclable materials.	
Micro	At a micro level, the replacement of the "take-make-dispose" business model implies a greater emphasis on new products, servicing, resource pooling and marketing concepts.	El as a tool to address bottlenecks in product durability and quality, in designing efficient products and "dynamic CE business models".	Target/Type:- Goods and services;	Government role is key regarding the creation of a CE, ensuring adequate regulatory frameworks, and encouraging the awareness of actors and social participation.	Albu, 2011; Bakker et al., 2014; Mont, 2008; Sanyé- Mengual et al., 2014.
			Mechanism:- Technological.	CE considerations may prove to be an opportunity for positive business differentiation, the development of new CE-friendly business models, and increasing resource efficiency.	

intended to be exhaustive but rather a testimony of the most "real world" relevance features identified in the corpus. It should be stressed that CE-inducing El policy and business strategy constitute a fertile ground to be investigated in more depth in the future.

6. Concluding remarks

What is the role of innovation in a sustainable techno-economic paradigm transition? This paper tried to address this question by focusing on the ways "eco-innovation" (EI) promotes a "circular economy" (CE). To clarify the CE-EI link, a corpus of specialised academic literature was identified and reviewed. The research comprehensively covered published peer-reviewed journal articles. By critically and analytically re-using this material, this paper strived to provide new insights regarding the connections between EI and the CE. In doing so it proposed the concept of "clean congruence" as a bridge between the two literature and sought to illuminate the *dimensions* that are more instrumental in achieving a CE at a variety of *levels*.

The overlap between the EI and CE literature is a fertile ground for fine-tuning the definitional issues that still remain open. Based on the literature, this paper advanced working definitions of CE and EI. CE is defined as an approach towards sustainable development achieved through several strategies aiming at: "input minimisation and efficient use of regenerative resources", i.e. material and energy efficiency and focus on renewable and non-hazardous materials; "life cycle extension and systems reconceptualization", designed from the outset for efficiency, reuse, repair, and recycling; "output reduction valorisation and waste minimisation" focused on resource and waste minimisation in production-consumption systems.

El is defined as a set of technological and non-technological innovations that prevent, mitigate, and allow recovery from environmental damage. Overall, technological El seems to enjoy wider popularity as a heuristic for the transition to a CE. However, the need to add other dimensions, beyond science and technology, is increasingly being recognised, with "softer" service and marketing innovations becoming emerging issues in the literature.

Regarding the connection between the concepts, the creation of a CE seems to be contingent on a process based on cooperation and multi-actor "systemic" integration, with El emerging as a pathway for achieving that. The El-CE research shows the importance of what has been emphasised generally as "clean congruence" at the macro-level, as "green collective innovation" at the meso level and "dynamic CE business models" at the micro level.

The methodological constraints of "meta" studies are the most relevant shortcomings of this paper, particularly related to randomisation and the representativeness of the sample. A literature review using bibliometric considerations implies inherent biases linked with keyword definitions and the limited number of sources (Li and Zhao, 2015). This paper did not collect material from books or reports, for example, in the assumption that journal articles are usually the preferred means for academics and practitioners to publish their newest research (Chappin and Ligtvoet, 2014; Linnenluecke and Griffiths, 2013; Markard et al., 2012). The findings could nevertheless benefit from contextualisation using these wider sources of data, particularly considering recent vibrant CE agendas in several organisations. The identification of search terms also carried implicit risks of exclusion and biases. For instance, the predominance of macro and meso perspectives in the corpus may be attributable to biases in the initial choice of keywords. However it may also be due to still-evolving definitions of CE being mostly focused, so far, on the macro level. These caveats underscore the necessity of critical reflection about the findings, as well as the need for further empirical research. This paper did not aim to exhaustively collect examples, but rather to express the core thinking behind the literature, so as to enable the emergence of new conclusions regarding the connection between CE and EI. Nonetheless, the methodological and database limitations may still be alleviated in future, by expanding the knowledge base from which lessons are drawn.

Finally, this is a dynamic field with a wide range of opportunities for further research. A deeper understanding of the connections between the CE and EI is still elusive, requiring more empirical methods for assessing and measuring their mutual influence, in particular regarding the role of EI in implementing a CE. Moreover, since these concepts have a wide application, the way in which they are used and understood by different stakeholders varies. As a result, a better understanding of these diverse perspectives is needed in order to better tailor strategies and policies. For example, the role of consumers as "part of the supply chain" and "innovative agents" in the development of a CE has not yet been properly addressed.

Acknowledgements

The first author is supported by the Portuguese Foundation for Science and Technology (FCT), under the grant SFRH/BD/52295/ 2013, and CENSE - Center for Environmental and Sustainability Research. CENSE is financed through the Strategic Project Pest-OE/ AMB/UI4085/2013 from the Portuguese Science and Technology Foundation. This work was also supported by the Portuguese Science and Technology Foundation Grant UID/GES/00315/2013 and Strategic Project (UID/ECO/00436/2013). The funding sources played no part in the design, analysis, interpretation, or writing of the article, or in the decision to publish. The authors would like to thank the five anonymous reviewers we were given the opportunity to interact with, as well as the editors for their valuable remarks and suggestions. Also we would like to acknowledge the insightful comments of Dan Ward.

References

- Adams, M., Ghaly, A.E., 2007. Maximizing sustainability of the Costa Rican coffee industry. J. Clean. Prod. Mater. Flow Anal. Mater. Flow Manag. 15, 1716–1729. https://doi.org/10.1016/j.jclepro.2006.08.013.
- Aghion, P., Hemous, D., Veugelers, R., 2009. No Green Growth without Innovation. Bruegel, Brussels. Bruegel Policy Brief 2009/07.
- Albu, A., 2011. Business innovation using industrial ecology. Metal. Int. 16, 28–31. Alvarez, S.A., Barney, J.B., McBride, R., Wuebker, R., 2014. Realism in the study of entrepreneurship. Acad. Manag. Rev. 39, 227–231. https://doi.org/10.5465/ amr.2013.0244
- Anastas, P.T., Lankey, R.L., 2000. Life cycle assessment and green chemistry: the yin and yang of industrial ecology. Green Chem. 2, 289–295. https://doi.org/ 10.1039/B005650M.
- Andersen, M.S., 2007. An introductory note on the environmental economics of the circular economy. Sustain Sci. 2, 133–140. https://doi.org/10.1007/s11625-006-0013-6.
- Andersen, M.M., 2008. Eco-innovation towards taxonomy and a theory. In: Presented at the Entrepreneurship and Innovation – Organizations, Institutions, Systems and Regions, DRUID Conference, Copenhagen, CBS, Denmark.
- Andrews, C., DeVault, D., 2009. Green Niche market development. J. Ind. Ecol. 13, 326–345. https://doi.org/10.1111/j.1530-9290.2009.00112.x.
- Ayres, R., Weaver, P. (Eds.), 1998. Eco-restructuring; Implications for Sustainable Development. United Nations University Press.
- Baas, L., 2011. Planning and uncovering industrial symbiosis: comparing the Rotterdam and Östergötland regions. Bus. Strat. Env. 20, 428–440. https://doi.org/ 10.1002/bse.735.
- Baas, L., Hjelm, O., 2015. Support your future today: enhancing sustainable transitions by experimenting at academic conferences. J. Clean. Prod. 98, 1–7. https:// doi.org/10.1016/j.jclepro.2015.02.059. Special Volume: Support your future today! Turn environmental challenges into opportunities.
- Bakker, C., Wang, F., Huisman, J., den Hollander, M., 2014. Products that go round: exploring product life extension through design. J. Clean. Prod. 69, 10–16. https://doi.org/10.1016/j.jclepro.2014.01.028.
- Bakshi, B.R., Ziv, G., Lepech, M.D., 2015. Techno-ecological synergy: a framework for sustainable engineering. Environ. Sci. Technol. 49, 1752–1760. https://doi.org/ 10.1021/es5041442.

- Balconi, M., Brusoni, S., Orsenigo, L., 2010. In defence of the linear model: an essay. Res. Policy 39, 1–13. https://doi.org/10.1016/j.respol.2009.09.013.
- Barberio, G., Buttol, P., Masoni, P., Scalbi, S., Andreola, F., Barbieri, L., Lancellotti, I., 2010. Use of incinerator bottom ash for frit production. J. Ind. Ecol. 14, 200–216. https://doi.org/10.1111/j.1530-9290.2010.00224.x.
- Barney, J.B., 2003. Resources, capabilities, core competencies, invisible assets, and knowledge assets: label proliferation and theory development in the field of strategic management. In: Helfat, C.E. (Ed.), The SMS Blackwell Handbook of Organizational Capabilities, pp. 422–426.
- Basu, A.J., van Zyl, D.J.A., 2006. Industrial ecology framework for achieving cleaner production in the mining and minerals industry. J. Clean. Prod. Improv. Environ. Econ. Ethical Perform. Min. Ind. 14, 299–304. https://doi.org/10.1016/j.jclepro.2004.10.008. Part 1. Environmental Management and Sustainable Development Improving Environmental, Economic and Ethical Performance in the Mining Industry. Part 1. Environmental Management and Sustainable Development.
- Bergek, A., Jacobsson, S., Carlsson, B., Lindmark, S., Rickne, A., 2008. Analyzing the functional dynamics of technological innovation systems: a scheme of analysis. Res. Policy 37, 407–429. https://doi.org/10.1016/j.respol.2007.12.003.
- Bergquist, A.-K., Söderholm, K., Kinneryd, H., Lindmark, M., Söderholm, P., 2013. Command-and-control revisited: environmental compliance and technological change in Swedish industry 1970–1990. Ecological Economics. New Clim. Econ. 85, 6–19. https://doi.org/10.1016/j.ecolecon.2012.10.007.
- Birat, J.-P., 2015. Life-cycle assessment, resource efficiency and recycling. Metall. Res. Technol. 112, 206. https://doi.org/10.1051/metal/2015009.
- Blowfield, M., Johnson, L., 2013. Turnaround Challenge: Business and the City of the Future. OUP, Oxford.
- Bocken, N.M.P., Short, S.W., Rana, P., Evans, S., 2014. A literature and practice review to develop sustainable business model archetypes. J. Clean. Prod. 65, 42–56. https://doi.org/10.1016/j.jclepro.2013.11.039.
- Boons, F., Lüdeke-Freund, F., 2013. Business models for sustainable innovation: state-of-the-art and steps towards a research agenda. J. Clean. Prod. 45, 9–19. https://doi.org/10.1016/j.jclepro.2012.07.007.
- Boons, F., Spekkink, W., Mouzakitis, Y., 2011. The dynamics of industrial symbiosis: a proposal for a conceptual framework based upon a comprehensive literature review. J. Clean. Prod. 19, 905–911. https://doi.org/10.1016/j.jclepro.2011.01.003.
- Boons, F., Montalvo, C., Quist, J., Wagner, M., 2013. Sustainable innovation, business models and economic performance: an overview. J. Clean. Prod. Sustain. Innovat. Bus. Models 45, 1–8. https://doi.org/10.1016/j.jclepro.2012.08.013.
- Boulding, K.E., 1966. The economics of the coming spaceship Earth. In: Jarrett, H. (Ed.), Environmental Quality in a Growing Economy: Essays from the Sixth RFF Forum. Routledge, pp. 3–15.
- Braungart, M., McDonough, W., 2002. Cradle to Cradle: Remaking the Way We Make Things, first ed. North Point Press, New York.
- Brils, J., de Boer, P., Mulder, J., de Boer, E., 2014. Reuse of dredged material as a way to tackle societal challenges. J. Soils Sediments 14, 1638–1641. https://doi.org/ 10.1007/s11368-014-0918-0.
- Bristow, G., Wells, P., 2005. Innovative discourse for sustainable local development: a critical analysis of eco-industrialism. Int. J. Innovat. Sustain. Dev. 1, 168. https://doi.org/10.1504/IJISD.2005.008086.
- Busch, J., Steinberger, J.K., Dawson, D.A., Purnell, P., Roelich, K., 2014. Managing critical materials with a technology-specific stocks and flows model. Environ. Sci. Technol. 48, 1298–1305. https://doi.org/10.1021/es404877u.
- Caraça, J., Lundvall, B.-Å., Mendonça, S., 2009. The changing role of science in the innovation process: from Queen to Cinderella? Technol. Forecast. Soc. Change 76, 861–867. https://doi.org/10.1016/j.techfore.2008.08.003.
- Carrillo-Hermosilla, J., Gonzaléz, P., del, R., Könnölä, T., 2009. Eco-innovation: when Sustainability and Competitiveness Shake Hands. Palgrave Macmillan.
- Carrillo-Hermosilla, J., del Río, P., Könnölä, T., 2010. Diversity of eco-innovations: reflections from selected case studies. J. Clean. Prod. 18, 1073–1083. https:// doi.org/10.1016/j.jclepro.2010.02.014.
- Castellaci, F., Grodal, S., Mendonca, S., Wibe, M., 2005. Advances and challenges in innovation studies. J. Econ. Issue 39, 91–121. https://doi.org/10.1080/ 00213624.2005.11506782.
- Cecelja, F., Raafat, T., Trokanas, N., Innes, S., Smith, M., Yang, A., Zorgios, Y., Korkofygas, A., Kokossis, A., 2015. e-Symbiosis: technology-enabled support for Industrial Symbiosis targeting Small and Medium Enterprises and innovation. J. Clean. Prod. 98, 336–352. https://doi.org/10.1016/j.jclepro.2014.08.051. Special Volume: Support your future today! Turn environmental challenges into opportunities.
- Cerón-Palma, I., Sanyé-Mengual, E., Oliver-Solà, J., Montero, J.-I., Rieradevall, J., 2012. Barriers and opportunities regarding the implementation of rooftop eco. Greenhouses (RTEG) in Mediterranean cities of Europe. J. Urban Technol. 19, 87–103. https://doi.org/10.1080/10630732.2012.717685.
- Ceschin, F., 2013. Sustainable Product-Service Systems: Between Strategic Design and Transition Studies. Springer Science & Business Media.
- Chappin, E.J.L., Ligtvoet, A., 2014. Transition and transformation: a bibliometric analysis of two scientific networks researching socio-technical change. Renew. Sustain. Energy Rev. 30, 715–723. https://doi.org/10.1016/j.rser.2013.11.013.
- Charter, M., Clark, T., 2007. Sustainable innovation key conclusions from sustainable innovation conferences 2003–2006 organised by the centre for sustainable design. Centre Sustain. Des.
- Chen, Y.J., Sheu, J.-B., 2013. Pursuing extended producer responsibility in the context of EIPs by a Hotelling model. J. Clean. Prod. 57, 152–157. https://doi.org/ 10.1016/j.jclepro.2013.05.025.

- Chen, X., Fujita, T., Ohnishi, S., Fujii, M., Geng, Y., 2012. The impact of scale, recycling boundary, and type of waste on symbiosis and recycling. J. Ind. Ecol. 16, 129–141. https://doi.org/10.1111/j.1530-9290.2011.00422.x.
- Cheng, C.Y., 2007. China's new development plan: strategy, Agenda, and prospects. Asian Aff. Am. Rev. 34, 47–59. https://doi.org/10.3200/AAFS.34.1.47-60.
- Chertow, M.R., 2007. "Uncovering" industrial symbiosis. J. Ind. Ecol. 11, 11–30. https://doi.org/10.1162/jiec.2007.1110.
- Chertow, M.R., Ehrenfeld, J., 2012. Organizing self-organizing systems. J. Ind. Ecol. 16, 13–27. https://doi.org/10.1111/j.1530-9290.2011.00450.x.
- Cohen, M.J., 2006. Ecological modernization and its discontents: the American environmental movement's resistance to an innovation-driven future. Futures 38, 528–547. https://doi.org/10.1016/i.futures.2005.09.002.
- Collado-Ruiz, D., Ostad-Ahmad-Ghorabi, H., 2013. Estimating environmental behavior without performing a life cycle assessment. J. Ind. Ecol. 17, 31–42. https://doi.org/10.1111/j.1530-9290.2012.00510.x.
- Corder, G.D., Golev, A., Giurco, D., 2015. "Wealth from metal waste": translating global knowledge on industrial ecology to metals recycling in Australia. Miner. Eng. Sustain. Miner. 76, 2–9. https://doi.org/10.1016/j.mineng.2014.11.004.
- Cuerva, M.C., Triguero-Cano, Á., Córcoles, D., 2014. Drivers of green and non-green innovation: empirical evidence in Low-Tech SMEs. J. Clean. Prod. 68, 104–113. https://doi.org/10.1016/j.jclepro.2013.10.049.
- Curran, T., Williams, I.D., 2012. A zero waste vision for industrial networks in Europe. In: Journal of Hazardous Materials, Selected Papers Presented at the 2nd International Conference CRETE 2010, October 2010-Industrial and Hazardous Waste Management, vol. 207, pp. 3–7. https://doi.org/10.1016/ j.jhazmat.2011.07.122.
- Dalziel, M., 2006. The impact of industry associations: evidence from statistics Canada data. Innovat. Manag. Policy Pract. 8, 296.
- de Jesus, A., Mendonça, S., 2018. Lost in Transition? Drivers and Barriers in the Ecoinnovation Road to the Circular Economy. Ecol. Econ 145, 75–89. https:// doi.org/10.1016/j.ecolecon.2017.08.001.
- del Río, P., Carrillo-Hermosilla, J., Könnölä, T., 2010. Policy strategies to promote ecoinnovation. J. Ind. Ecol. 14, 541–557. https://doi.org/10.1111/j.1530-9290.2010.00259.x.
- Desrochers, P., 2000. Market processes and the closing of 'industrial loops'. J. Ind. Ecol. 4, 29–43. https://doi.org/10.1162/108819800569276.
- Desrochers, P., 2004. Industrial symbiosis: the case for market coordination. J. Clean. Prod. Appl. Ind. Ecol. 12, 1099–1110. https://doi.org/10.1016/ j.jclepro.2004.02.008.
- Deutz, P., 2009. Producer responsibility in a sustainable development context: ecological modernisation or industrial ecology? Geogr. J. 175, 274–285. https:// doi.org/10.1111/j.1475-4959.2009.00330.x.
- Dewick, P., Foster, C., Green, K., 2007. Technological Change and the Environmental Impact of Food Production and Consumption (SSRN Scholarly Paper No. ID 1005183). Social Science Research Network, Rochester, NY.
- Dienst, C., Schneider, C., Xia, C., Saurat, M., Fischer, T., Vallentin, D., 2013. On track to become a low carbon future City? First findings of the integrated status quo and trends assessment of the pilot city of Wuxi in China. Sustainability 5, 3224–3243. https://doi.org/10.3390/su5083224.
- Dobbs, R., Brinkman, M., Jeremy, O., Fraser, Thompson, F., Zornes, M., 2011. Resource Revolution: Meeting the World's Energy, Materials, Food, and Water Needs. McKinsey & Company.
- Dong, L, Fujita, T., Zhang, H., Dai, M., Fujii, M., Ohnishi, S., Geng, Y., Liu, Z., 2013a. Promoting low-carbon city through industrial symbiosis: a case in China by applying HPIMO model. Energy Policy 61, 864–873. https://doi.org/10.1016/ j.enpol.2013.06.084.
- Dong, L., Zhang, H., Fujita, T., Ohnishi, S., Li, H., Fujii, M., Dong, H., 2013b. Environmental and economic gains of industrial symbiosis for Chinese iron/steel industry: Kawasaki's experience and practice in Liuzhou and Jinan. J. Clean. Prod. 59, 226–238. https://doi.org/10.1016/j.jclepro.2013.06.048.
- Dong, L., Gu, F., Fujita, T., Hayashi, Y., Gao, J., 2014. Uncovering opportunity of lowcarbon city promotion with industrial system innovation: case study on industrial symbiosis projects in China. Energy Policy 65, 388–397. https://doi.org/ 10.1016/j.enpol.2013.10.019.
- EC, 2011a. Innovation for a Sustainable Future the Eco-innovation Action Plan (Eco-AP), COM(2011) 899 Final. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions. European Commission.
- EC, 2011b. Horizon 2020-The Framework Programme for Research and Innovation, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions.
- EC, 2011c. A Resource-efficient Europe Flagship Initiative under the Europe 2020 Strategy (Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions No. COM(2011) 21). European Commission.
- EC, 2012. The World and Europe up to 2050: EU Policies and Research Priorities Global Europe 2050. European Commission.
- EC, 2014. Progress Report on the Roadmap to a Resource Efficient Europe. (Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions No. SWD(2014) 206 Final), Accompanying the Document towards a Circular Economy: a Zero Waste Programme for Europe. European Commission.
- EC, 2015. Closing the Loop an EU Action Plan for the Circular Economy (Communication from the Commission to the European Parliament, the

Council, the European Economic and Social Committee and the Committee of the Regions. No. COM(2015) 614 Final). European Commission, Directorate-General for the Environment.

- EC, 2016. Horizon 2020 Work Programme 2016-2017 17. Cross-cutting Activities (Focus Areas). European Commission.
- EC, 2017. Report on the Implementation of the Circular Economy Action Plan (Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions. No. COM(2017) 33 Final). European Commission, Directorate-General for the Environment.
- EIO, 2011. The Eco-innovation Challenge: Pathways to a Resource-efficient Europe. Eco-innovation Observatory. Funded by the European Commission, DG Environment, Brussels.
- ElO, 2013. The Eco-innovation Gap: an Economic Opportunity for Business. Ecoinnovation Observatory. Funded by the European Commission, DG Environment, Brussels.
- EIO, 2016. Policies and Practices for Eco-innovation Uptake and Circular Economy Transition. Eco-innovation Observatory. Funded by the European Commission, DG Environment, Brussels.
- Elsevier, 2014. Scopus [WWW Document]. http://www.elsevier.com/online-tools/ scopus.
- EMF, 2012. Towards the Circular Economy: Economic and Business Rationale for an Accelerated Transition (No. Vol. 1). Ellen MacArthur Foundation (EMF).
- EMF, 2013. Towards the Circular Economy: Opportunities for the Consumer Goods Sector (No. Vol. 2). Ellen MacArthur Foundation (EMF).
- EMF, 2014. Towards the Circular Economy: Accelerating the Scale-up across Global Supply Chains (No. Vol. 3). Ellen MacArthur Foundation (EMF); World Economic Forum. McKinsey & Company.
- Erdmann, L., Hilty, LM., 2010. Scenario analysis. J. Ind. Ecol. 14, 826–843. https:// doi.org/10.1111/j.1530-9290.2010.00277.x.
- Fagerberg, J., Mowery, D.C., Nelson, R.R., 2004. The Oxford Handbook of Innovation, New Edition Edition. Oxford University Press, Oxford; New York.
- Falagas, M.E., Pitsouni, E.I., Malietzis, G.A., Pappas, G., 2008. Comparison of PubMed, Scopus, Web of science, and google scholar: strengths and weaknesses. FASEB J. 22, 338–342. https://doi.org/10.1096/fj.07-9492LSF.
- Fichter, K., Hintemann, R., 2014. Beyond energy. J. Ind. Ecol. 18, 846–858. https:// doi.org/10.1111/jiec.12155.
- Fiksel, J., 2002. Sustainable development through industrial ecology. Advancing sustainability through green chemistry and engineering. ACS Symp. Ser. 823, 13–29.
- Finster, M., Eagan, P., Hussey, D., 2001. Linking industrial ecology with business strategy: creating value for green product design. J. Ind. Ecol. 5, 107–125. https://doi.org/10.1162/108819801760049495.
- Franceschet, M., 2009. A comparison of bibliometric indicators for computer science scholars and journals on web of science and google scholar. Scientometrics 83, 243–258. https://doi.org/10.1007/s11192-009-0021-2.
- Freeman, C., Louçã, F., 2001. As time goes By : From the Industrial Revolutions to the Information Revolution: from the Industrial Revolutions to the Information Revolution. Oxford University Press.
- Freeman, C., Soete, L., 1997. The Economics of Industrial Innovation. Pinter, London. Frosch, R.A., Gallopoulos, N.E., 1989. Strategies for manufacturing. Sci. Am. 261, 144–152.
- Ganapathy, S.P., Natarajan, J., Gunasekaran, A., Subramanian, N., 2014. Influence of eco-innovation on Indian manufacturing sector sustainable performance. Int. J. Sustain. Dev. World Ecol. 21, 198–209. https://doi.org/10.1080/ 13504509.2014.907832.
- Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res. Policy 31, 1257–1274. https://doi.org/10.1016/S0048-7333(02)00062-8.
- Geels, F.W., 2011. The multi-level perspective on sustainability transitions: responses to seven criticisms. Environ. Innov. Soc. Transit 1, 24–40. https:// doi.org/10.1016/j.eist.2011.02.002.
- Geissdoerfer, M., Savaget, P., Bocken, N.M.P., Hultink, E.J., 2017. The Circular Economy – a new sustainability paradigm? J. Clean. Prod. 143, 757–768. https:// doi.org/10.1016/j.jclepro.2016.12.048.
- Geng, Y., Doberstein, B., 2008. Greening government procurement in developing countries: building capacity in China. J. Environ. Manag. 88, 932–938. https:// doi.org/10.1016/j.jenvman.2007.04.016.
- Geng, Y., Zhang, P., Côté, R.P., Fujita, T., 2009a. Assessment of the national ecoindustrial park standard for promoting industrial symbiosis in China. J. Ind. Ecol. 13, 15–26. https://doi.org/10.1111/j.1530-9290.2008.00071.x.
- Geng, Y., Zhu, Q., Doberstein, B., Fujita, T., 2009b. Implementing China's circular economy concept at the regional level: a review of progress in Dalian, China. Waste Manag, 29, 996–1002. https://doi.org/10.1016/j.wasman.2008.06.036.
- Geng, Y., Tsuyoshi, F., Chen, X., 2010a. Evaluation of innovative municipal solid waste management through urban symbiosis: a case study of Kawasaki. J. Clean. Prod. 18, 993–1000. https://doi.org/10.1016/j.jclepro.2010.03.003.
- Geng, Y., Xinbei, W., Qinghua, Z., Hengxin, Z., 2010b. Regional initiatives on promoting cleaner production in China: a case of Liaoning. J. Clean. Prod. 18, 1502–1508. https://doi.org/10.1016/j.jclepro.2010.06.028.
- Geng, Y., Fu, J., Sarkis, J., Xue, B., 2012. Towards a national circular economy indicator system in China: an evaluation and critical analysis. J. Clean. Prod. 23, 216–224. https://doi.org/10.1016/j.jclepro.2011.07.005.
- Geng, Y., Liu, Z., Xue, B., Dong, H., Fujita, T., Chiu, A., 2014. Emergy-based assessment on industrial symbiosis: a case of shenyang economic and technological

development zone. Environ. Sci. Pollut. Res. 21, 13572-13587. https://doi.org/ 10.1007/s11356-014-3287-8.

- Georgesçu-Roegen, N., 1971. The Entropy Law and the Economic Process. Harvard University Press.
- Ghisellini, P., Cialani, C., Ulgiati, S., 2016. A review on circular economy: the expected transition to a balanced interplay of environmental and economic systems. J. Clean. Prod. Towards Post Fossil Carbon Soc. Regen. Prev. Eco-Ind. Dev. 114, 11–32. https://doi.org/10.1016/j.jclepro.2015.09.007.
- Giannetti, B.F., Bonilla, S.H., Almeida, C.M.V.B., 2004. Developing eco-technologies: a possibility to minimize environmental impact in Southern Brazil. J. Clean. Prod. 12, 361–368. https://doi.org/10.1016/S0959-6526(03)00033-7.
- Graedel, T.E., Cao, J., 2010. Metal spectra as indicators of development. PNAS 107, 20905–20910. https://doi.org/10.1073/pnas.1011019107.
- Grundmann, V., Bilitewski, B., Zehm, A., Darbra, R.M., Barceló, D., 2013. Risk-based management of chemicals and products in a circular economy at a global scale-Impacts of the FP7 funded project RISKCYCLE. Environ. Sci. Eur. 25, 14. https:// doi.org/10.1186/2190-4715-25-14.
- Grupo Interministerial Economia Circular, 2017. Liderar a transição [plano de ação para a economia circular em Portugal: 2017-2020]. Ministério do Ambiente; Ministério da Economia, Ministério da Ciência, Tecnologia e Ensino Superior.
- Guan, J., Liu, N., 2016. Exploitative and exploratory innovations in knowledge network and collaboration network: a patent analysis in the technological field of nano-energy. Res. Policy 45, 97–112. https://doi.org/10.1016/ i.respol.2015.08.002.
- Guide, V.D.R., van Wassenhove, L.N., 2009. OR FORUM—the evolution of closedloop supply chain research. Oper. Res. 57, 10–18. https://doi.org/10.1287/ opre.1080.0628.
- Gupta, S., Palsule-Desai, O.D., 2011. Sustainable supply chain management: review and research opportunities. IIMB Manag. Rev. 23, 234–245. https://doi.org/ 10.1016/j.iimb.2011.09.002.
- Heiskanen, E., Lovio, R., 2010. User–Producer interaction in housing energy innovations. J. Ind. Ecol. 14, 91–102. https://doi.org/10.1111/j.1530-9290.2009.00196.x.
- Hewes, A.K., Lyons, D.I., 2008. The humanistic side of eco-industrial parks: champions and the role of trust. Reg. Stud. 42, 1329–1342. https://doi.org/10.1080/ 00343400701654079.
- Heyes, A., Kapur, S., 2011. Regulatory attitudes and environmental innovation in a model combining internal and external R&D. J. Environ. Econ. Manag. 61, 327–340. https://doi.org/10.1016/j.jeem.2010.12.003.
- Huber, J., 2000. Towards industrial ecology: sustainable development as a concept of ecological modernization. J. Environ. Policy Plann 2, 269–285. https:// doi.org/10.1002/1522-7200(200010/12)2:4<269::AID-JEPP58>3.0.C0;2-U.
- Huesemann, M.H., Huesemann, J.A., 2007. Will progress in science and technology avert or accelerate global collapse? A critical analysis and policy recommendations. Environ. Dev. Sustain 10, 787–825. https://doi.org/10.1007/s10668-007-9085-4.
- IAU, 2013. Économie circulaire, écologie industrielle Éléments de réflexion à l'échelle de l'Île-de-Franc, L'Institut d'aménagement et d'urbanisme de la région d'Île-de-France.
- Jabbour, C.J.C., Neto, A.S., Gobbo Jr., J.A., Ribeiro, M., de, S., Jabbour, A.B.L., de, S., 2015. "Eco-innovations in more sustainable supply chains for a low-carbon economy: a multiple case study of human critical success factors in Brazilian leading companies. Int. J. Prod. Econ. 164, 245–257. https://doi.org/10.1016/ j.ijpe.2014.11.015.
- Jacobsen, N.B., 2006. Industrial symbiosis in Kalundborg, Denmark: a quantitative assessment of economic and environmental aspects. J. Ind. Ecol. 10, 239–255. https://doi.org/10.1162/108819806775545411.
- Ji, G., Gunasekaran, A., Yang, G., 2014. Constructing sustainable supply chain under double environmental medium regulations. Int. J. Prod. Econ. Build. Supply Chain Syst. Capab. Age Glob. Complex. Emerg. Theor. Pract. 147, 211–219. https://doi.org/10.1016/j.ijpe.2013.04.012. Part B.
- Jin, Y., Wang, D., Wei, F., 2004. The ecological perspective in chemical engineering. Chemical Engineering Science. Complex Syst. Multi-scale Methodol. 59, 1885–1895. https://doi.org/10.1016/j.ces.2004.01.042.
- Jones, P.T., Geysen, D., Tielemans, Y., van Passel, S., Pontikes, Y., Blanpain, B., Quaghebeur, M., Hoekstra, N., 2013. Enhanced Landfill Mining in view of multiple resource recovery: a critical review. J. Clean. Prod. 55, 45–55. https:// doi.org/10.1016/j.jclepro.2012.05.021.
- Kemp, R., 2010. Eco-innovation: definition, measurement and open research issues. Econ. Polit. 397–420.
- Kemp, R., Schot, J., Hoogma, R., 1998. Regime shifts to sustainability through processes of niche formation: the approach of strategic niche management. Technol. Anal. Strateg. Manag. 10, 175–198. https://doi.org/10.1080/ 09537329808524310.
- Khan, K.S., Kunz, R., Kleijnen, J., Antes, G., 2003. Five steps to conducting a systematic review. J. R. Soc. Med. 96, 118–121.
- Killerby, S., Maplesden, F., Jack, M., McDonald, G., Rolland, D., 2007. Chemical modification of timber decking : assessing the parameters of acceptability. N. Z. J. For. Sci. 37, 412–434.
- Kirchherr, J., Reike, D., Hekkert, M., 2017. Conceptualizing the circular economy: an analysis of 114 definitions. Resour. Conserv. Recycl 127, 221–232. https:// doi.org/10.1016/j.resconrec.2017.09.005.
- Koenig, H.E., Cantlon, J.E., 1999. Quantitative industrial ecology & ecological economics. J. Ind. Ecol. 3, 63–83. https://doi.org/10.1162/108819899569557.
- Köhler, A.R., Hilty, L.M., Bakker, C., 2011. Prospective impacts of electronic textiles on

3016

recycling and disposal. J. Ind. Ecol. 15, 496-511. https://doi.org/10.1111/j.1530-9290.2011.00358.x.

- Körhönen, J., 2008. Reconsidering the economics logic of ecological modernization. Environ. Plan. A 40, 1331–1346. https://doi.org/10.1068/a38363.
- Körhönen, J., von Malmborg, F., Strachan, P.A., Ehrenfeld, J.R., 2004. Management and policy aspects of industrial ecology: an emerging research agenda. Bus. Strat. Env. 13, 289–305. https://doi.org/10.1002/bse.415.
- Lai, K., Wong, C.W.Y., Venus Lun, Y.H., 2014. The role of customer integration in extended producer responsibility: a study of Chinese export manufacturers. Int. J. Prod. Econ. 147, 284–293. https://doi.org/10.1016/j.ijpe.2013.06.028. Part B.
- Lee, Y.-N., Walsh, J.P., 2016. Inventing while you work: knowledge, non-R&D learning and innovation. Res. Policy 45, 345–359. https://doi.org/10.1016/ j.respol.2015.09.009.
- Lemke, F., Luzio, J.P.P., 2014. Exploring green consumers' mind-set toward green product design and life cycle assessment. J. Ind. Ecol. 18, 619–630. https:// doi.org/10.1111/jiec.12123.
- Levänen, J., 2015. Ending waste by law: institutions and collective learning in the development of industrial recycling in Finland. J. Clean. Prod. 87, 542–549. https://doi.org/10.1016/j.jclepro.2014.09.085.
- Li, W., Zhao, Y., 2015. Bibliometric analysis of global environmental assessment research in a 20-year period. Environ. Impact Assess. Rev. 50, 158–166. https:// doi.org/10.1016/j.eiar.2014.09.012.
- Lieder, M., Rashid, A., 2016. Towards circular economy implementation: a comprehensive review in context of manufacturing industry. J. Clean. Prod. 115, 36–51. https://doi.org/10.1016/j.jclepro.2015.12.042.
- Linnenluecke, M.K., Griffiths, A., 2013. Firms and sustainability: mapping the intellectual origins and structure of the corporate sustainability field. Glob. Environ. Change 23, 382–391. https://doi.org/10.1016/j.gloenvcha.2012.07.007.
- Liu, D., Li, H., Wang, W., Dong, Y., 2012. Constructivism scenario evolutionary analysis of zero emission regional planning: a case of Qaidam Circular Economy Pilot area in China. Int. J. Prod. Econ. 140, 341–356. https://doi.org/10.1016/ j.ijpe.2011.04.008.
- Lombardi, D.R., Laybourn, P., 2012. Redefining industrial symbiosis. J. Ind. Ecol. 16, 28–37. https://doi.org/10.1111/j.1530-9290.2011.00444.x.
- Lovins, A.B., Lovins, L.H., Hawken, P., 1999. Natural Capitalism: the Next Industrial Revolution. Earthscan.
- Markard, J., Raven, R., Truffer, B., 2012. Sustainability transitions: an emerging field of research and its prospects. Res. Policy, Special Sect. Sustain. Transit. 41, 955–967. https://doi.org/10.1016/j.respol.2012.02.013.
- Mathews, J.A., Tan, H., 2011. Progress toward a circular economy in China. J. Ind. Ecol. 15, 435–457. https://doi.org/10.1111/j.1530-9290.2011.00332.x.
 Matos, S., Hall, J., 2007. Integrating sustainable development in the supply chain:
- Matos, S., Hall, J., 2007. Integrating sustainable development in the supply chain: the case of life cycle assessment in oil and gas and agricultural biotechnology. J. Oper. Manag. Supply Chain Manag. Sustain. Environ. Special Issue Front. Empir. Supply Chain Res. 25, 1083–1102. https://doi.org/10.1016/ j.jom.2007.01.013.
- Mattinen, M.K., Nissinen, A., Hyysalo, S., Juntunen, J.K., 2015. Energy use and greenhouse gas emissions of air-source heat pump and innovative groundsource air heat pump in a cold climate. J. Ind. Ecol. 19, 61–70. https://doi.org/ 10.1111/jiec.12166.
- Matus, K.J.M., Xiao, X., Zimmerman, J.B., 2012. Green chemistry and green engineering in China: drivers, policies and barriers to innovation. J. Clean. Prod. 32, 193–203. https://doi.org/10.1016/j.jclepro.2012.03.033.
- Maurizio Catulli, E.F., 2012. Information and communication technology-enabled low carbon technologies a new subsector of the economy? J. Ind. Ecol. 16, 1088–1980. https://doi.org/10.1111/j.1530-9290.2011.00452.x.
- Mendonça, S., 2017. On the discontinuity of the future by other means: reviewing the foresight world of Richard Slaughter. Futures 86, 84–91. https://doi.org/ 10.1016/j.futures.2016.08.005.
- Mirabella, N., Castellani, V., Sala, S., 2014. Current options for the valorization of food manufacturing waste: a review. J. Clean. Prod. 65, 28–41. https://doi.org/ 10.1016/j.jclepro.2013.10.051.
- Mirata, M., Emtairah, T., 2005. Industrial symbiosis networks and the contribution to environmental innovation: the case of the Landskrona industrial symbiosis programme. J. Clean. Prod. 10th Anniv. Int. Inst. Ind. Environ. Econ. (IIIEE) Manag. policy Sustain. Consum. Prod. 13, 993–1002. https://doi.org/10.1016/ j.jclepro.2004.12.010.
- Mirhedayatian, S.M., Azadi, M., Farzipoor Saen, R., 2014. A novel network data envelopment analysis model for evaluating green supply chain management. Int. J. Prod. Econ. Build. Supply Chain Syst. Capab. Age Glob. Complex. Emerg. Theor. Pract. 147, 544–554. https://doi.org/10.1016/j.ijpe.2013.02.009. Part B.
- Mont, O., 2008. Innovative approaches to optimising design and use of durable consumer goods. Int. J. Prod. Dev. 6, 227–250. https://doi.org/10.1504/ IJPD.2008.020395.
- Moreno, A., Cappellaro, F., Masoni, P., Amato, A., 2011. Application of product data technology standards to LCA data. J. Ind. Ecol. 15, 483–495. https://doi.org/ 10.1111/j.1530-9290.2011.00353.x.
- Moya-Anegón, F., de, Chinchilla-Rodríguez, Z., Vargas-Quesada, B., Corera-Álvarez, E., Muñoz-Fernández, F.J., González-Molina, A., Herrero-Solana, V., 2007. Coverage analysis of Scopus: a journal metric approach. Scientometrics 73, 53–78. https://doi.org/10.1007/s11192-007-1681-4.
- Muñoz, I., Milà i Canals, L., Clift, R., 2008. Consider a spherical man. J. Ind. Ecol. 12, 521–538. https://doi.org/10.1111/j.1530-9290.2008.00060.x.
- Nelson, R.R., Winter, S.G., 1982. An Evolutionary Theory of Economic Change. Belknap Press, Cambridge, Massachusetts and London, England.

- Nguyen, Q.C., Ye, F., 2015. Study and evaluation on sustainable industrial development in the Mekong Delta of Vietnam. J. Clean. Prod. https://doi.org/10.1016/ j.jclepro.2014.08.087.
- Niza, S., Rosado, L., Ferrão, P., 2009. Urban metabolism. J. Ind. Ecol. 13, 384–405. https://doi.org/10.1111/j.1530-9290.2009.00130.x.
- OECD, 2009. Sustainable Manufacturing and Eco-innovation. Framework, Practices and Measurement. Organisation for Economic Co-operation and Development, Paris. Synthesis report.
- OECD, 2010. Eco-innovation in Industry: Enabling Green Growth. Organisation for Economic Co-operation and Development, Paris.
- OECD, 2012. OECD Environmental Outlook to 2050. Organisation for Economic Cooperation and Development, Paris.
- OECD/Eurostat, 2005. Oslo Manual Guidelines for Collecting and Interpreting Innovation, the Measurement of Scientific and Technological Activities. OECD Publishing, Paris.
- Ogunseitan, O.A., 2007. Public health and environmental benefits of adopting leadfree solders. JOM 59, 12–17. https://doi.org/10.1007/s11837-007-0082-8.
- Paquin, R.L., Howard-Grenville, J., 2012. The evolution of facilitated industrial symbiosis. J. Ind. Ecol. 16, 83–93. https://doi.org/10.1111/j.1530-9290.2011.00437.x.
- Park, J.Y., Chertow, M.R., 2014. Establishing and testing the "reuse potential" indicator for managing wastes as resources. J. Environ. Manag. 137, 45–53. https:// doi.org/10.1016/j.jenvman.2013.11.053.
- Park, H.-S., Rene, E.R., Choi, S.-M., Chiu, A.S.F., 2008. Strategies for sustainable development of industrial park in Ulsan, South Korea—from spontaneous evolution to systematic expansion of industrial symbiosis. J. Environ. Manag. 87, 1–13. https://doi.org/10.1016/j.jenvman.2006.12.045.
- Park, J., Sarkis, J., Wu, Z., 2010. Creating integrated business and environmental value within the context of China's circular economy and ecological modernization. J. Clean. Prod. 18, 1494–1501. https://doi.org/10.1016/ j.jclepro.2010.06.001.
- Patala, S., Hämäläinen, S., Jalkala, A., Pesonen, H.-L., 2014. Towards a broader perspective on the forms of eco-industrial networks. J. Clean. Prod. 82, 166–178. https://doi.org/10.1016/j.jclepro.2014.06.059.
- Patnaik, R., Poyyamoli, G., 2015. Developing an eco-industrial park in Puducherry region, India – a SWOT analysis. J. Environ. Plan. Manag. 58, 976–996. https:// doi.org/10.1080/09640568.2014.904768.
- Pauli, G., 1997. Zero emissions: the ultimate goal of cleaner production. J. Clean. Prod. Ind. Ecol. 5, 109–113. https://doi.org/10.1016/S0959-6526(97)00013-9.
- Pauli, G., 2010. Blue Economy-10 Years, 100 Innovations, 100 Million Jobs. Paradigm Pubns, Taos, NM.
- Pearce, D.W., Turner, R.K., 1990. Economics of Natural Resources and the Environment. Harvester Wheatsheaf.
- Potočnik, J., 2014. Eco-innovation and circular economy. In: Presented at the Opening of 16th European Forum on Eco-innovation, European Commission.
- Preston, F., 2012. Briefing paper a global Redesign? Shaping the circular economy. Energy, Environ. Resour. Gov.
- Ruiz Puente, M.C., Arozamena, E.R., Evans, S., 2015. Industrial symbiosis opportunities for small and medium sized enterprises: preliminary study in the Besaya region (Cantabria, Northern Spain). J. Clean. Prod. 87, 357–374. https://doi.org/ 10.1016/j.jclepro.2014.10.046.
- Raafat, T., Trokanas, N., Cecelja, F., Bimi, X., 2013. An ontological approach towards enabling processing technologies participation in industrial symbiosis. In: Computers & Chemical Engineering, Selected Papers from ESCAPE-22 (European Symposium on Computer Aided Process Engineering - 22), 17-20 June 2012, London, UK, vol. 59, pp. 33–46. https://doi.org/10.1016/ j.compchemeng.2013.03.022.
- Raafat, T., Trokanas, N., Cecelja, F., Kokossis, A., Yang, A., 2012. Semantically-enabled formalisation to support and automate the application of industrial symbiosis. Comput. Aided Chem. Eng. 31 https://doi.org/10.1016/B978-0-444-59506-5.50042-0.
- Rashid, A., Asif, F.M.A., Krajnik, P., Nicolescu, C.M., 2013. Resource Conservative Manufacturing: an essential change in business and technology paradigm for sustainable manufacturing. J. Clean. Prod. 57, 166–177. https://doi.org/10.1016/ j.jclepro.2013.06.012.
- Reh, L., 2013. Process engineering in circular economy. Particuology. Meas. Technol. Part. Syst. 11, 119–133. https://doi.org/10.1016/j.partic.2012.11.001.
- Rennings, K., 2000. Redefining innovation eco-innovation research and the contribution from ecological economics. Ecol. Econ. 32, 319–332. https:// doi.org/10.1016/S0921-8009(99)00112-3.
- Riding, M.J., Herbert, B.M.J., Ricketts, L., Dodd, I., Ostle, N., Semple, K.T., 2015. Harmonising conflicts between science, regulation, perception and environmental impact: the case of soil conditioners from bioenergy. Environ. Int. 75, 52–67. https://doi.org/10.1016/j.envint.2014.10.025.
- Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F.S.I., Lambin, E., Lenton, T., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., de Wit, C., Hughes, T., van der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R., Fabry, V., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P., Foley, J., 2009. Planetary boundaries: exploring the safe operating space for humanity. Ecol. Soc. 14 https://doi.org/ 10.5751/ES-03180-140232.
- Sanyé-Mengual, E., Pérez-López, P., González-García, S., Lozano, R.G., Feijoo, G., Moreira, M.T., Gabarrell, X., Rieradevall, J., 2014. Eco-designing the use phase of products in sustainable manufacturing. J. Ind. Ecol. 18, 545–557. https://doi.org/ 10.1111/jiec.12161.

- Sauvé, S., Bernard, S., Sloan, P., 2016. Environmental sciences, sustainable development and circular economy: alternative concepts for trans-disciplinary research. Environ. Dev. 17, 48–56. https://doi.org/10.1016/j.envdev.2015.09.002.
- Schot, J., Kanger, L., 2016. Deep Transitions: Emergence, Acceleration, Stabilization and Directionality (SPRU Working Paper Series No. 2016–15). SPRU - Science and Technology Policy Research, University of Sussex.
- Schot, J., Steinmueller, E., 2016. Framing Innovation Policy for Transformative Change: Innovation Policy 3.0. Ser.- Sci. Policy Res. Unit SPRU Univ. Sussex. Work, Pap.
- Schulte, U.G., 2013. New business models for a radical change in resource efficiency. Environ. Innovat. Soc. Transit. Energy Mater. Growth Homage Robert Ayres 9, 43–47. https://doi.org/10.1016/j.eist.2013.09.006.
- Schumpeter, J., 1928. The instability of capitalism. Econ. J. 38, 361–386. https:// doi.org/10.2307/2224315.
- Shi, L., Yu, B., 2014. Eco-industrial parks from strategic niches to development mainstream: the cases of China. Sustainability 6, 6325–6331. https://doi.org/ 10.3390/su6096325.
- Shi, H., Chertow, M.R., Song, Y., 2010. Developing country experience with ecoindustrial parks: a case study of the Tianjin Economic-Technological Development Area in China. J. Clean. Prod. 18, 191–199. https://doi.org/10.1016/ j.jclepro.2009.10.002.
- Short, S.W., Bocken, N.M.P., Barlow, C.Y., Chertow, M.R., 2014. From refining sugar to growing tomatoes. J. Ind. Ecol. 18, 603–618. https://doi.org/10.1111/jiec.12171.
- Silva, A.J.O., Mendonça, S., 2015. The grand green challenge: assessing progress in eco-innovation through Y02 patents. In: Global Sustainable Development Report, 2015 Edition. United Nations - Department of Economic and Social Affairs, UN, New York.
- Silva, D.A.L., de Oliveira, J.A., Saavedra, Y.M.B., Ometto, A.R., Rieradevall i Pons, J., Gabarrell Durany, X., 2015. Combined MFA and LCA approach to evaluate the metabolism of service polygons: a case study on a university campus. Resour. Conservat. Recycl. 94, 157–168. https://doi.org/10.1016/j.resconrec.2014.11.001.
- Simboli, A., Taddeo, R., Morgante, A., 2014. Analysing the development of Industrial Symbiosis in a motorcycle local industrial network: the role of contextual factors. J. Clean. Prod. 66, 372–383. https://doi.org/10.1016/j.jclepro.2013.11.045.
- Smith, A., Voß, J.-P., Grin, J., 2010. Innovation studies and sustainability transitions: the allure of the multi-level perspective and its challenges. Res. Policy, Special Sect. Innovat. Sustain. Transit. 39, 435–448. https://doi.org/10.1016/ j.respol.2010.01.023.
- Soete, L., 2013. Is innovation always good? In: Fagerberg, J., Martin, B.R., Andersen, E.S. (Eds.), Innovation Studies :: Evolution and Future Challenges. Oxford Univ. Press, pp. 134–144.
- Stahel, W.R., 1982. Product-life factor. In: Mitchell Prize Winning Paper 1982.
- Stahel, W.R., 1997. The Functional Economy: Cultural and Organizational Change. In: Richards, D.J. (Ed.), The Industrial Green Game: Implications for Environmental Design and Management. Washington, DC, pp. 91–100.
- Stahel, W.R., 2010. The Performance Economy, second ed. Palgrave Macmillan, Basingstoke. First Edition 2006.
- Stahel, W.R., Reday-Mulvey, G., 1981. Jobs for Tomorrow: the Potential for Substituting Manpower for Energy. Vantage Press, New York.
- Sterr, T., Ott, T., 2004. The industrial region as a promising unit for eco-industrial development—reflections, practical experience and establishment of innovative instruments to support industrial ecology. J. Clean. Prod. Appl. Ind. Ecol. 12, 947–965. https://doi.org/10.1016/j.jclepro.2004.02.029.
- Su, B., Heshmati, A., Geng, Y., Yu, X., 2013. A review of the circular economy in China: moving from rhetoric to implementation. J. Clean. Prod. 42, 215–227. https://doi.org/10.1016/j.jclepro.2012.11.020.
- Thomas, V., Theis, T., Lifset, R., Grasso, D., Kim, B., Koshland, C., Pfahl, R., 2003. Industrial ecology: policy potential and research needs. Environ. Eng. Sci. 20.
- Tombesi, P., 2006. Good thinking and poor value: on the socialization of knowledge in construction. Build. Res. Inf. 34, 272–286. https://doi.org/10.1080/ 09613210600634336.
- Tregner-Mlinaric, A., Repo, P., 2014. The Eco-innovation Action Plan in an Environmental Policy Context (CASI EU Policy Brief No2 No. CASI Policy Brief 30.6.2014 CASI-PB-2.2014 EUROPE).
- UN, 2015. Global Sustainable Development Report, 2015 Edition. United Nations -Department of Economic and Social Affairs.
- UNFCCC, 2016. 175 Parties sign Paris Climate Agreement [WWW Document]. United Nations Framework Convention on Climate Change. http://newsroom.unfccc. int/media/632121/list-of-representatives-to-high-level-signature-ceremony. pdf (Accessed 16 November 2017).
- UNEP, 2006. Circular Economy: an Alternative Model for Economic Development. United Nations Environment Programme.
- UNEP, 2011. Decoupling Natural Resource Use and Environmental Impacts from Economic Growth, UNEP, International Resource Panel. United Nations Environment Programme.

- UNEP, 2014. Circular Economy. United Nations Environment Programme (UNEP) logo United Nations Environment Programme. UNEP, UNECE, 2016. GEO-6 Assessment for the pan-European Region, (Rev. 1).
- United Nations Environment Programme (UNEP), Nairobi, Kenya, van Berkel, R., Fujita, T., Hashimoto, S., Geng, Y., 2009. Industrial and urban sym-
- biosis in Japan: analysis of the Eco-Town program 1997–2006. J. Environ. Manag. 90, 1544–1556. https://doi.org/10.1016/j.jenvman.2008.11.010.
- van den Bergh, J.C.J.M., 2013. Robert Ayres, ecological economics and industrial ecology. Environ. Innovat. Soc. Transit. Energy Mater. Growth Homage Robert Ayres 9, 1–7. https://doi.org/10.1016/j.eist.2013.09.008.
- van den Bergh, J.C.J.M., Truffer, B., Kallis, G., 2011. Environmental innovation and societal transitions: introduction and overview. Environ. Innovat. Soc. Transit. 1, 1–23. https://doi.org/10.1016/j.eist.2011.04.010.
- Vanner, R., Bicket, M., Withana, S., ten Brink, P., Razzini, P., van Dijl, E., Watkins, E., Hestin, M., Tan, A., Guilcher, S., Hudson, C., 2014. Scoping Study to Identify Potential Circular Economy Actions, Priority Sectors, Material Flows & Value Chains (DG Environment's Framework Contract for Economic Analysis ENV.F.1/ FRA/2010/0044 No. Final Report). Policy Studies Institute (PSI), Institute for European Environmental Policy (IEEP), BIO and Ecologic Institute.
 Vivanco, D., Kemp, R., van der Voet, E., Heijungs, R., 2014. Using LCA-based
- Vivanco, D., Kemp, R., van der Voet, E., Heijungs, R., 2014. Using LCA-based decomposition analysis to study the multidimensional contribution of technological innovation to environmental pressures. J. Ind. Ecol. 18, 380–392. https://doi.org/10.1111/jiec.12118.
- Von Hippel, E., 2005. Democratizing Innovation. MIT Press, Cambridge London.
- Watkins, G., Husgafvel, R., Pajunen, N., Dahl, O., Heiskanen, K., 2013. Overcoming institutional barriers in the development of novel process industry residue based symbiosis products – case study at the EU level. Miner. Eng. 41, 31–40. https://doi.org/10.1016/j.mineng.2012.10.003.
- Watkins, A., Papaioannou, T., Mugwagwa, J., Kale, D., 2015. National innovation systems and the intermediary role of industry associations in building institutional capacities for innovation in developing countries: a critical review of the literature. Res. Policy 44, 1407–1418. https://doi.org/10.1016/ j.respol.2015.05.004.
- Weber, K.M., Hemmelskamp, J., 2005. Towards Environmental Innovation Systems. Springer Science & Business Media.
- Wen, C.F., Zhao, Y.L., Liang, R.Z., 2007. Recycle of low chemical potential substance. Res. Conservat. Recycl. 51, 475–486. https://doi.org/10.1016/ j.resconrec.2006.10.011.
- Wijkman, A., Skånberg, 2015. The Circular Economy and Benefits for Society. Interim report by the Club of Rome with support from the MAVA Foundation and the Swedish Association of Recycling Industries.
- WWF, 2014. Living Planet Report 2014-Species and Spaces, People and Places. World Wide Fund for Nature.
- Xue, B., Chen, X., Geng, Y., Guo, X., Lu, Cheng-peng, Zhang, Z., Lu, Chen-yu, 2010. Survey of officials' awareness on circular economy development in China: based on municipal and county level. Res. Conservat. Recycl. 54, 1296–1302. https:// doi.org/10.1016/j.resconrec.2010.05.010.
- Yarime, M., 2007. Promoting green innovation or prolonging the existing technology. J. Ind. Ecol. 11, 117–139. https://doi.org/10.1162/jiec.2007.1151.
- Yu, C., de Jong, M., Dijkema, G.P.J., 2014. Process analysis of eco-industrial park development – the case of Tianjin, China. J. Clean. Prod. 64, 464–477. https:// doi.org/10.1016/j.jclepro.2013.09.002.
- Yu, F., Han, F., Cui, Z., 2015. Evolution of industrial symbiosis in an eco-industrial park in China. J. Clean. Prod. 87, 339–347. https://doi.org/10.1016/ j.jclepro.2014.10.058.
- Yuan, Z., Bi, J., Moriguichi, Y., 2006. The circular economy: a new development strategy in China. J. Ind. Ecol. 10, 4–8. https://doi.org/10.1162/ 108819806775545321.
- Zaman, A.U., 2015. A comprehensive review of the development of zero waste management: lessons learned and guidelines. J. Clean. Prod. 91, 12–25. https:// doi.org/10.1016/j.jclepro.2014.12.013.
- Zhijun, F., Nailing, Y., 2007. Putting a circular economy into practice in China. Sustain. Sci. 2, 95–101. https://doi.org/10.1007/s11625-006-0018-1.
- Zhu, Q., Geng, Y., 2013. Drivers and barriers of extended supply chain practices for energy saving and emission reduction among Chinese manufacturers. J. Clean. Prod. 40, 6–12. https://doi.org/10.1016/j.jclepro.2010.09.017.
- Zhu, Q., Geng, Y., Lai, K., 2010. Circular economy practices among Chinese manufacturers varying in environmental-oriented supply chain cooperation and the performance implications. J. Environ. Manage 91, 1324–1331. https://doi.org/10.1016/j.jenvman.2010.02.013.
- Zhu, Q., Geng, Y., Sarkis, J., Lai, K.-H., 2015. Barriers to promoting eco-industrial parks development in China. J. Ind. Ecol. 19, 457–467. https://doi.org/10.1111/ iiec.12176.
- ZWIA, 2009. Zero Waste Definition [WWW Document]. Zero Waste International Alliance (Accessed 7 April 2016).